

## 

**Citation:** Otero I, Castellnou M, González I, Arilla E, Castell L, Castellví J, et al. (2018) Democratizing wildfire strategies. Do you realize what it means? Insights from a participatory process in the Montseny region (Catalonia, Spain). PLoS ONE 13 (10): e0204806. https://doi.org/10.1371/journal. pone.0204806

Editor: Asko Noormets, Texas A&M University, UNITED STATES

Received: September 27, 2017

Accepted: September 14, 2018

Published: October 16, 2018

**Copyright:** © 2018 Otero et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper.

**Funding:** This research was funded by the German Excellence Initiative through the Integrative Research Institute on Transformations of Human-Environment Systems (IRI THESys) of the Humboldt University of Berlin. The public participation sessions were partly funded by the Spanish government's project CSO2014-54513-R (SINALECO). The funders had no role in study

RESEARCH ARTICLE

## Democratizing wildfire strategies. Do you realize what it means? Insights from a participatory process in the Montseny region (Catalonia, Spain)

lago Otero <sup>1,2</sup>\*, Marc Castellnou<sup>3</sup>°, Itziar González<sup>4</sup>°, Etel Arilla<sup>3</sup>, Llorenç Castell<sup>3‡</sup>, Jordi Castellví<sup>3‡</sup>, Francesc Sánchez<sup>4</sup>, Jonas Ø. Nielsen<sup>1,5</sup>

1 Integrative Research Institute on Transformations of Human-Environment Systems, Humboldt-Universität zu Berlin, Berlin, Germany, **2** Institute of Environmental Science and Technology, Universitat Autònoma de Barcelona, Cerdanyola del Vallès, Spain, **3** Grup de Reforç d'Actuacions Forestals (GRAF), Fire Department, Department of Home Affairs, Catalan Regional Government, Cerdanyola del Vallès, Spain, **4** Institut Cartogràfic de la Revolta, Barcelona, Spain, **5** Geography Department, Humboldt-Universität zu Berlin, Berlin, Germany

• These authors contributed equally to this work.

‡ These authors also contributed equally to this work.

\* iago.otero.armengol@hu-berlin.de

## Abstract

Participatory planning networks made of government agencies, stakeholders, citizens and scientists are receiving attention as a potential pathway to build resilient landscapes in the face of increased wildfire impacts due to suppression policies and land-use and climate changes. A key challenge for these networks lies in incorporating local knowledge and social values about landscape into operational wildfire management strategies. As large wildfires overcome the suppression capacity of the fire departments, such strategies entail difficult decisions about intervention priorities among different regions, values and socioeconomic interests. Therefore there is increasing interest in developing tools that facilitate decisionmaking during emergencies. In this paper we present a method to democratize wildfire strategies by incorporating social values about landscape in both suppression and prevention planning. We do so by reporting and critically reflecting on the experience from a pilot participatory process conducted in a region of Catalonia (Spain). There, we built a network of researchers, practitioners and citizens across spatial and governance scales. We combined knowledge on expected wildfires, landscape co-valuation by relevant actors, and citizen participation sessions to design a wildfire strategy that minimized the loss of social values. Drawing on insights from political ecology and transformation science, we discuss what the attempt to democratize wildfire strategies entails in terms of power relationships and potential for social-ecological transformation. Based on our experience, we suggest a trade-off between current wildfire risk levels and democratic management in the fire-prone regions of many western countries. In turn, the political negotiation about the landscape effects of wildfire expert knowledge is shown as a potential transformation pathway towards lower risk landscapes that can re-define agency over landscape and foster community re-learning on



design, data collection and analysis, decision to publish, or preparation of the manuscript. There was no additional external funding received for this study.

**Competing interests:** The authors have declared that no competing interests exist.

fire. We conclude that democratizing wildfire strategies ultimately entails co-shaping the landscapes and societies of the future.

## 1. Introduction

As climate change, land-use changes and suppression policies exacerbate the wildfire problem in many western countries, concern emerges on how to coexist with this perturbation [1]. Among the different options debated in the literature, collaborative or participatory planning networks made of public agencies, stakeholders and citizens are receiving attention as a potential tool to build resilience to wildfires [2–7]. Related to this endeavour is the notion of community-based fire management, where local communities plan their own wildfire regimes through partnerships with government agencies, NGOs and the private sector [8]. The development of these networks is driven both by a normative motivation to democratize wildfire management [9] and by the recognition that the complexity of the wildfire problem requires new governance arrangements operating at multiple scales [2,10,11]. This is in line with studies on adaptive governance showing that developing the right links between institutions across multiple organizational levels is crucial to build social-ecological resilience [12–15].

A key challenge for participatory wildfire planning networks lies in incorporating landscape values, local knowledge, and social perceptions of risk into operational decision making systems [9,16–18]. There is a large range of values that might be affected by wildfire events and management [19]. These heterogeneous values are embedded in often conflicting visions and policies of wildfire risk reduction [20]. Moreover, in many western countries local or traditional fire knowledge systems vanished with industrialization [21], and nowadays local communities have very limited experience in fire management as compared to expert agencies [1]. Which and whose values and knowledge are included in participatory wildfire planning networks are crucial questions when trying to build resilient landscapes. New scientific, technical and deliberative methodologies are therefore required, and our paper addresses this need.

Recently some practitioners have stressed the need to open the wildfire strategies of the fire departments up to democratic decision-making. 'Strategy' refers to the set of responses applied by the fire department to reduce the uncertainty created by a wildfire, involving a prioritization of the interventions to achieve concrete objectives such as, for instance, confining the wildfire within 150 ha [22,23]. Developing a strategy requires considering the wildfire potential, its suppression opportunities, and the people, land-uses, and properties at risk, as well as making decisions about the wildfire impacts on these factors [22]. As large wildfires increasingly overcome the suppression capacity of the fire departments [24–26], strategies entail difficult decisions about intervention priorities among different areas, landscape values and socioeconomic interests. Therefore there is increasing interest in developing tools to include the values and interests at stake into operational wildfire strategies, in order to facilitate decision-making processes during emergencies [27]. While 'strategy' mostly refers to the management of wildfires when they occur, it can be planned in advance by anticipating the wildfire patterns of a certain region and planning preventive landscape interventions. Thus, in this paper 'strategy' encompasses both prevention and suppression practices.

In La Jonquera large wildfire (Catalonia, Spain, 2012), one of us (M.C., GRAF's analyst) opened up the strategy of the Fire Department by reaching an agreement on intervention priorities with politicians, firefighters and stakeholders of the affected landscapes during an improvised meeting. Since then similar approaches were used during all significant wildfires

occurring in Catalonia. Based on that first experience, an attempt was made to plan participatory strategies with stakeholders without the pressure of the emergency, in Eastern Mourns (Northern Ireland, UK). In theory, the knowledge about wildfire patterns of a given region can provide scenarios to decision making, and planning the strategies in advance has the potential to reduce uncertainty during emergencies.

However, operationalizing this call is far from easy. As we show in this paper, the attempt to democratize wildfire strategies not only adds complexity to the challenge of developing participatory planning networks which are sensitive to social values about landscape, but it has also dramatic implications in terms of power relationships, agency over landscape and the potential for social-ecological transformation in fire-prone regions. We present a method to democratically plan wildfire strategies by reporting and critically reflecting on the experience from a pilot participatory process conducted in the Montseny region (Catalonia, Spain). Drawing on insights from political ecology [28–30], transformation science [31–34] and action-research [35,36] we argue that democratizing wildfire strategies ultimately entails co-shaping the societies and landscapes of the future.

The paper is structured as follows. Section 2 describes the study region. Section 3 explains the participatory process for the democratization of wildfire strategies which combines knowledge on expected wildfires, landscape co-valuation by relevant actors, and citizen participation sessions. In section 4 we critically reflect on the process based on an analysis of I.O.'s project diary and other documents such as minutes of meetings and presentations. Section 5 discusses the implications of our experience and section 6 concludes.

## 2. Context: The Montseny region

### 2.1. General features

A region of 60,596 ha was selected in the Barcelona metropolitan region, belonging to the autonomous region of Catalonia in north-eastern Spain (Fig 1A and 1B). The region includes the pre-littoral ranges of Montseny–reaching ca. 1700 m a.s.l.–and the littoral ranges of Montnegre-Corredor–reaching ca. 760 m a.s.l. (Fig 1C). Between the two ranges lay the large plain of Granollers, drained by the Mogent river, and the narrow plain of Sant Celoni, along which the Tordera river flows. The highway AP-7 and the high-speed train–connecting the Spanish and French coasts–pass through these plains, which host a large share of the region's population as well as most of its industrial activities (Fig 1C and 1D). The coastal strip of the region has a highway and hosts several touristic towns. Settlement is structured in towns or cities, sprawled residential developments, and isolated farmhouses. The city of Granollers with ca. 60,000 inhabitants, and several cities between 15,000 and 20,000 inhabitants such as Cardedeu, Sant Celoni or Arenys de Mar are placed in the region. The inland municipalities belong to the Vallès Oriental county and the coastal ones belong to the Maresme county.

The region has gone through a dramatic rural-to-urban socioeconomic transformation during the 20<sup>th</sup> century. Cropland and pastureland has shrunk while forests, urban and industrial areas have expanded. This has led to substantive changes in landscape ecology and biodiversity [37] and to a very vulnerable landscape concerning wildfires [38]. Both mountain ranges are partly protected by natural parks, the Montseny one being a UNESCO Biosphere Reserve. All these changes have gone hand in hand with an erosion of local fire knowledge, especially in the plateaus of Montseny where the traditional fires performed by shepherds for pasture regeneration were banned by the Natural Park in the early 1980s [39]. Forestland ownership in our study region is mostly private, even if some estates within the natural parks are owned by a public agency (the Barcelona Province Authority, in charge of managing the natural parks).



Fig 1. Presentation of the study region. a) Catalonia within Spain and Europe. b) Location of the Montseny region within the Barcelona province, in Catalonia. c) Main territorial characteristics of the Montseny region. d) Orthophoto of the Montseny region (2015). Settlement, industrial activities and strategic transport infrastructures in extensively forested mountain ranges make the region highly vulnerable to large wildfires. Source: own elaboration with data from Institut Cartogràfic i Geològic de Catalunya and Instituto Geográfico Nacional. Note: "Cardadeu" is misspelled. The correct spelling is "Cardedeu".

https://doi.org/10.1371/journal.pone.0204806.g001

PLOS ONE

## 2.2. Wildfire pattern

The wildfire pattern of the region is mostly composed by convection dominated wildfires with wind, but it also includes topographic wildfires [40,41]. In convection dominated wildfires, the wildfire's convective column transmits heat and throws spots to the surroundings thereby accelerating combustion. Wind increases spotting distance, creating new ignitions outside the convective column's influence zone and further accelerating wildfire spread. High fuel availability in the forests, stemming from historical landscape changes, determines high wildfire intensity. With these conditions, the wildfire quickly overcomes the Fire Department's suppression capacity. The largest historical wildfire in the region responded to this pattern. This



**Fig 2. Wildfire pattern and participatory dynamics in the Montseny region.** The region was divided in wildfire contention polygons (purple lines) and suggested strategic management points (light green areas). A method for landscape co-valuation was tested in polygons A and B (red lines). Landscape co-valuation and citizen participation sessions were conducted at the pilot level in polygons 1 to 5. The towns hosting citizen participation sessions (Montseny and Sant Esteve de Palautordera) are indicated. The 1994 large wildfire's perimeter is likewise shown. Source: own elaboration with data from Institut Cartogràfic i Geològic de Catalunya, Instituto Geográfico Nacional and Catalan Fire Department. Note: "Cardadeu" is misspelled. The correct spelling is "Cardedeu".

https://doi.org/10.1371/journal.pone.0204806.g002

PLOS ONE

wildfire resulted from the union of two wildfires that started in Gualba and Santa Coloma de Farners on the 10<sup>th</sup> August 1994. They united two days later, affecting about 11,000 hectares (Fig 2). In that event, west winds, local winds, accumulated drought, very high temperatures,

relative humidity below 30% and fuel continuity combined to produce an out-of-control wildfire [42]. Topographic wildfires are instead driven by local topographic winds and follow the slopes with the steepest gradient and highest insolation; they also include wildfires driven by the turn of the sea breezes [41]. In worst case scenarios, in our study region the Fire Department expects simultaneous large wildfires with virulent behaviour causing civil emergencies, as they threaten residential areas and key economic activities and transport routes (Fig 1C and 1D).

## 2.3. Wildfire governance system

The wildfire governance system in the region is composed of several institutions operating at different scales, whose interactions drive management outcomes on the ground. The town councils implement municipal wildfire prevention schemes with the support of the Technical Office of Municipal Wildfire Prevention of the Barcelona Province Authority (TOMWP) and in collaboration with the forest defence associations. The latter are heterogeneous municipal or multi-municipal groups of forest landowners, town councils and volunteers working with prevention and auxiliary extinction. Forest defence associations federate at the county level. During a wildfire, they are under the command of the Fire Department, which deals with the suppression operations and the protection of people.

The Fire Department belongs to the Department of Home Affairs of the Catalan regional government and operates at the Catalan scale. It is organized in seven emergency regions. GRAF is the Fire Department's wildfire specialists group, and has a technician in each emergency region. In the Northern Metropolitan Emergency Region where our study region belongs, GRAF's technician plans low fuel strips in collaboration with the TOMWP, forest landowners and the Forest Property Centre. Forest landowners form associations at the scale of mountain range (one in Montseny and one in Montnegre-Corredor). These associations defend the interests of private forest owners before public agencies. The one in Montnegre-Corredor develops forest planning for wildfire prevention with the support of the TOMWP. The Forest Property Centre is a public agency enhancing forest planning in private forests across Catalonia through the approval of estate and municipal schemes, often with wildfire prevention as a criterion. The natural parks are managed by the Barcelona Province Authority and have their own wildfire prevention schemes. These mostly consist in the management of forest tracks and water infrastructures. Finally, the county offices of the Barcelona Territorial Services of the Catalan Department of Agriculture implement general directives from the Department's Services of Wildfire Prevention and Forest Management.

# 3. The participatory process: A method for the democratization of wildfire strategies

## 3.1. Motivations

The method for the democratization of wildfire strategies evolved out of the synergies between the motivations of the core team (Table 1). These motivations ranged from a theoretical interest in understanding the social-ecological reconfigurations needed to coexist with wildfire to a political commitment towards better public services, more democratic politics and social-ecological transformation. The participatory process was specifically part of the postdoctoral project "The political ecology of wildfires" developed by I.O. at IRI THESys [1,43]. It was conducted in the region where he used to live during the project's fieldwork, as an action-research project towards alternative, less flammable landscapes [36,44].

As the head of GRAF, M.C. is in charge of developing the strategies followed by the Fire Department during wildfires across Catalonia. In talks to wildfire experts and practitioners he

# 

Table 1. Composition and motivations of the core team of the participatory process.	The roles performed by each institution are shown in <u>S1 Table</u> . Source: own
elaboration.	

Person	Institution	Motivations
Iago Otero	Integrative Research Institute on Transformations of Human- Environment Systems (IRI THESys), Humboldt University of Berlin (Germany)	<ol> <li>Understand how social groups reconfigure their relations with the environment as they learn to coexist with wildfire.</li> <li>Contribute to initiate radical transformations towards alternative landscapes that are unable to burn at high intensity.</li> <li>Link wildfire prevention to alternative forest and land management practices embedded in regional networks of production and consumption.</li> </ol>
Marc Castellnou	Support Group for Forest Interventions (GRAF), Fire Department, Department of Home Affairs, Catalan regional government (Spain)	<ol> <li>Democratize the strategies of the Fire Department to prevent, suppress and manage wildfires, moving from technical to social strategies.</li> <li>Include the conflicting social values about landscape in the design of wildfire strategies, moving from the minimization of area burnt to the minimization of value loss.</li> <li>Increase the efficiency of the public service in the management of wildfire emergencies by introducing the common good in front of private interests.</li> </ol>
Itziar González	Association Cartographic Institute of Revolt (ICR), Barcelona (Spain)	<ol> <li>Overcome barriers for inter-agency and citizen-agency cooperation and trust, including non-transparency, corruption, and non-accountability.</li> <li>Empower citizens in the stewardship of common resources through participatory learning processes.</li> <li>Man emerging practices and networks of democratic politics</li> </ol>

https://doi.org/10.1371/journal.pone.0204806.t001

had previously stressed the need to democratize those strategies; something he thought should be done by including the often conflicting social values about landscape in their design. He aimed at doing so in this participatory process. The latter ambition was intended to strengthen the quality and accountability of the public service, in the face of financial cuts and potential privatization after the economic and debt crisis. Indeed, in Catalonia and Spain, the latter intermingled with a crisis of democratic institutions, with corruption eroding the trust of citizens, who mobilized in new practices of "real democracy" [45]. In this context, the main motivation of I.G. and her association was to empower citizens in the stewardship of common resources through renewed mechanisms of accountability of the public administration.

In the next sub-sections we summarize the main steps of the participatory process (see also Fig 3). A detailed timeline including the main tasks and the responsible institutions can be



**Fig 3. Participatory process for the democratization of wildfire strategies.** Rectangles show the main steps, and blue arrows show the (dis)connection between them. The green oval outlines the potential concrete outcomes of the process. Underlying the participatory process, there is a potential transformative pathway towards a new social-ecological setup. See text for details. Source: own elaboration.

https://doi.org/10.1371/journal.pone.0204806.g003

found in the supporting information (<u>S1 Table</u>). In what follows, 'facilitators' refers to IRI THESys and ICR (<u>Table 1</u>).

# 3.2. Mapping wildfire contention polygons and strategic management points based on the expected wildfire pattern

We started by drawing wildfire contention polygons in a Geographic Information System (GIS) (Figs 2 and 3). These were based in GRAF's knowledge of the wildfire pattern, stemming from a study of past wildfires [40]. In our study region, expected wildfires include topographic wildfires and convection dominated wildfires with wind [41]. Wildfire contention polygons typically follow crest lines and other topographic elements where wildfires change behaviour. They are used by GRAF analysts to assess wildfires' spreading pattern and potential size according to meteorological, topographic and fuel conditions, as well as to the effects of the implemented strategies. Essentially, they are areas within which a wildfire might be contained thus avoiding its propagation to neighbouring polygons.

Wildfire contention polygons were used as the basis of a participatory landscape valuation that would inform about the social priorities–i.e. which polygons were most valued and why– in order to translate them into concrete strategies. Strategies include strategic management points (SMP), which are wildfire friendly landscape structures that facilitate suppression and limit wildfire size (Fig 2). They are normally planned along crest lines and in valley bottoms, thus fragmenting the landscape and making fire spread more difficult.

## 3.3. Involving the actors of the wildfire governance system and codesigning the landscape valuation method

We identified the wildfire governance system, i.e. actors with competencies in wildfire prevention, wildfire extinction and forest management operating in our study region (Tables 2 and 3; Fig 3). This identification was done by drawing on the first author experience in wildfire governance projects across Catalonia. Their main scale of intervention ranged from Catalan to municipal. Most of the actors operated at intermediate scales, defined by administrative (province, county) or landscape (mountain ranges) boundaries.

A first group of actors from the wildfire governance system was selected including those operating at scales comprised between the autonomous region of Catalonia and the county. Group 1 actors included public agencies, associations of forest landowners, and public-private entities such as a federation of forest defence associations (Table 3). Between March and September 2015 individual meetings were held with these actors to present the project and involve them. A first joint meeting occurred in October 2015. In that meeting, the goals of the project and the methods of landscape valuation were discussed. As a follow-up of the meeting, we did a landscape valuation test in 2 polygons (1 in Montnegre-Corredor mountain range and 1 in Montseny mountain range; polygons A and B in Fig 2). For this, we asked actors to provide a text describing their interests and/or law-mandated competencies, including landscape values, wildfire and forest management tasks, challenges and proposals. They were asked to provide information in a way that would help citizens appreciate the specific value of the polygons during the participatory sessions that would be organized in the future.

We did a synthesis of the contributions and discussed it with the actors during a second meeting held in January 2016. We coded the synthesis of polygon A and created six categories of landscape values (Biodiversity and cultural heritage, Current socioeconomic activities, Cooperative fabric, Elements in need of special protection during wildfire; Potential for social and ecological economy; and Other values). Based on these categories, a sheet was designed to gather polygon-specific input on landscape values from actors (Table 4). The landscape

Table 2. Representation of the wildfire governance system of our study region, including the main actors and their scale of intervention. This representation was done for the purposes of the present participatory process, and it necessarily involved the inclusion of some actors and the exclusion of others (see Section 4.2). Source: own elaboration.

Scale of intervention	Actor(s)
Autonomous region of Catalonia	GRAF FPC
Barcelona province	TOMWP
Northern Barcelona Metropolitan Region	NMER
Montseny mountain ranges	MAFL NPM
Montnegre-Corredor mountain ranges	MCAFL PMC
County of Vallès Oriental	VOCO FFDAVO
County of Maresme	MCO FFDAM
Multiple municipalities	FDAs
Municipality	TOMWP Town Councils FDAs

GRAF: Support Group for Forest Interventions, Fire Department, Catalan Department of Home Affairs.

FPC: Forest Property Centre, Catalan Department of Agriculture.

TOMWP: Technical Office of Municipal Wildfire Prevention, Barcelona Province Authority.

NMER: Northern Metropolitan Emergency Region, Fire Department, Catalan Department of Home Affairs.

MAFL: Montseny Association of Forest Landowners.

NPM: Natural Park of Montseny, Barcelona Province Authority.

MCAFL: Montnegre-Corredor Association of Forest Landowners.

PMC: Park of Montnegre-Corredor, Barcelona Province Authority.

VOCO: Vallès Oriental County Office, Barcelona Territorial Service, Catalan Department of Agriculture.

FFDAVO: Federation of Forest Defence Associations Vallès Oriental County.

MCO: Maresme County Office, Barcelona Territorial Service, Catalan Department of Agriculture.

FFDAM: Federation of Forest Defence Associations Maresme County.

FDAs: Forest Defence Associations.

https://doi.org/10.1371/journal.pone.0204806.t002

valuation design was supported by the creation of a GIS composed of the layers used by the different actors in their wildfire and forest management tasks (Fig 4).

A second group of actors (Group 2, Table 3) was selected that could provide complementary information on diverse landscape values ranging from biodiversity and cultural heritage to land management options. They had mostly a regional focus and included museums, universities and an environmentalist platform. Representatives from these actors were invited to a joint meeting in January 2016 where the project was presented.

## 3.4 Mapping alternative forest and land management projects

A map of alternative land-use projects was started by I.O. and his commune's housemates, interested in building a network of people with experience in agro-ecological land reclamation practices (Fig 3). These practices were considered to have the potential to influence a shift towards less flammable landscapes through forestry, grazing and agriculture. At the same time, they could contribute to re-frame the wildfire issue as a matter of transformation of the current land-use model, characterized by the abandonment of forest and land management in mountain ranges. This map was expected to converge with the landscape valuation and the GIS



Table 3. Actors participating in the process. In meetings, each actor was characteristically represented by 1–2 persons. Besides the representative/s, other members of the institutions were regularly informed of the process' developments. Group 3 actors included the town mayor and/or some town councillors. In some cases the mayor represented both the town council and the local forest defence association. Source: own elaboration.

Actor	Scale of intervention	Reasons for inclusion in the process		
ROUP 1: Actors of the wildfire governance system operating at scales comprised between the autonomous region of Catalonia and the county				
Support Group for Forest Interventions (GRAF), Fire Department, Catalan Department of Home Affairs	Autonomous region of Catalonia	Before a wildfire, it plans prevention works and forecasts risk. During the wildfire, it develops the strategy as decision support to the emergency's head.		
Forest Property Centre, Catalan Department of Agriculture	Autonomous region of Catalonia	Public administration that enhances forest planning and management in private forests by means of schemes at the estate and municipal levels. Their guidelines for forest management integrate the prevention of large wildfires as a key objective.		
Technical Office of Municipal Wildfire Prevention, Barcelona Province Authority	Municipalities of the Barcelona province	It drafts wildfire prevention schemes for municipalities, including forest tracks, water infrastructures and the protection of residential areas. These schemes are developed in cooperation with the town councils and the forest defence associations. It also develops forest planning for wildfire prevention in private estates in collaboration with the Montnegre-Corredor Association of Forest Landowners.		
Northern Metropolitan Emergency Region, Fire Department, Catalan Department of Home Affairs	Northern Barcelona Metropolitan Region	It manages the resources of the region's fire stations, coordinates prevention tasks, plans activities related to risk monitoring, and commands the operations during wildfire suppression.		
Montseny Association of Forest Landowners	Montseny mountain ranges	It groups together Montseny forest landowners. It aims at uniting efforts, looking for new agricultural and forest management opportunities, and disseminating the socio-environmental benefits of sustainable resource management in Montseny.		
Natural Park of Montseny, Barcelona Province Authority	Montseny mountain ranges	It manages a significant part of the study area with the aim of making compatible the conservation of natural, landscape and cultural values with socioeconomic development and public use. It manages forest tracks as part of its own wildfire prevention scheme. The Natural Park is a UNESCO Biosphere Reserve.		
Montnegre-Corredor Association of Forest Landowners	Montnegre-Corredor mountain ranges	It acts as a representative of Montnegre-Corredor forest landowners before governmental agencies. It aims at enhancing the profitability of forestry and at recovering human population in the mountain farmhouses, abandoned after rural outmigration.		
Park of Montnegre-Corredor, Barcelona Province Authority	Montnegre-Corredor mountain ranges	It manages a significant part of the study area with the aim of making compatible the conservation of landscape values with socioeconomic development. It manages forest tracks and water infrastructures as part of its own wildfire prevention scheme.		
Vallès Oriental County Office and Maresme County Office, Barcelona Territorial Service, Catalan Department of Agriculture	Counties of Vallès Oriental and Maresme	They implement general directives from the Department's Wildfire Prevention Service and Forest Management Service. They authorize forestry works in those estates that do not have a planning scheme approved by the Forest Property Centre.		
Federation of Forest Defence Associations Vallès Oriental County	County of Vallès Oriental	It coordinates the activities of the county's forest defence associations (see Group 3, this table).		
GROUP 2: Actors with a regional focus providing comp	lementary information on landscape value	s		
Coordination Group for Montseny Defence	Montseny and Montnegre-Corredor mountain ranges	Environmentalist platform leading the social movement for the protection of Montseny since the 1980s. It provided polygon-specific information on landscape ecological and cultural values, mainly on the need to recover human population in the mountain after decades of land abandonment.		
Museum of Natural Sciences in Granollers	County of Vallès Oriental and mountain ranges of Montseny and Montnegre-Corredor	It monitors biodiversity in the study region. It provided polygon- specific information on the state of small mammals, bats, and butterflies, including habitats and species of special conservation interest.		
Department of Food and Animal Science, Autonomous University of Barcelona	Multiple	It develops research on plant ecological responses to grazing and on silvopastoral management alternatives. It provided polygon-specific information on vegetation diversity and proposals to recover open habitats linked to grazing activities in mountains.		

(Continued)

# 

#### Table 3. (Continued)

Actor	Scale of intervention	Reasons for inclusion in the process
Montseny Ethnology Museum in Arbúcies	Montseny mountain ranges	Centre of exhibition, conservation, dissemination and research on Montseny's cultural heritage. It provided polygon-specific information on archaeological and architectonic elements, as well as on ancient trees.
Centre for Ecological Research and Forestry Applications, and University of Barcelona	Multiple	It develops research on ecosystem structure and functioning; carbon and water balances; and ecophysiological responses of forests to climate change, droughts, wildfires and management. It provided general guidelines on how to manage forests to adapt to climate change, droughts and wildfires.
BeWater Project, Centre for Ecological Research and Forestry Applications	Tordera river catchment	Project promoting science-society collaboration for sustainable water management and adaptation to the impacts of global change in the Mediterranean. The Tordera is one of the river basins studied in the project. It provided general information on the relationship between forest cover and water resources, as well as silvopasture management alternatives.
Forest Museum	Mountain ranges of Montseny and Montnegre-Corredor	Project to create a forest museum in the town of Sant Celoni around the social-environmental heritage of forests. It did not provide information.
GROUP 3: Actors of the wildfire governance system op	erating in the 5 selected polygons for pilot v	aluation (municipal scale)
Town Council of Sant Esteve de Palautordera	Municipality of Sant Esteve de Palautordera	Municipalities included in the 5 pilot polygons. Town councils develop municipal wildfire prevention schemes with the support of
Town Council of Montseny	Municipality of Montseny	the Technical Office of Municipal Wildfire Prevention of the
Town Council of Sant Pere de Vilamajor	Municipality of Sant Pere de Vilamajor	Barcelona Province Authority and in collaboration with the forest defence associations. Schemes include: maintenance of forest tracks
Town Council of Fogars de Montclús	Municipality of Fogars de Montclús	and water infrastructures, protection of residential areas, surveillance, and emergency planning.
Forest Defence Association of Sant Esteve de Palautordera	Municipality of Sant Esteve de Palautordera	Forest defence associations covering the municipalities included in the 5 pilot polygons. Forest defence associations are made up of forest
Forest Defence Association of Sant Pere de Vilamajor, Sant Antoni de Vilamajor and Cardedeu	Municipalities of Sant Pere de Vilamajor, Sant Antoni de Vilamajor and Cardedeu	landowners, town councils and volunteers. They can cover one or more municipalities. They are engaged in prevention activities, which they develop in collaboration with the town councils, as well as in
Forest Defence Association of Montseny-Migjorn	Municipalities of Montseny, Fogars de Montclús and Campins	auxiliary extinction activities.

https://doi.org/10.1371/journal.pone.0204806.t003

developed with the actors. 28 projects were selected 1) working in the region's forests, croplands and pasturelands, 2) selling or buying products resulting from these, and 3) working towards the revitalization of rural activities and the enhancement of regional networks of production and consumption. An interview guide was designed to gather basic information about the projects and the kind of transformation they wanted to contribute to, as well as about the main challenges and suggested solutions. Four of them were visited and their promoters interviewed, including producers of artisan cheese, organic meat and firewood. However, the mapping was aborted due to fading of initial enthusiasm and lack of human resources.

## 3.5. Co-valuing landscape in a pilot area

Given the complexity of conducting a landscape co-valuation in the entire study region, we selected five contiguous polygons in the Montseny range to conduct a pilot citizen valuation (Fig 2, polygons 1–5; Fig 3). Actors from groups 1 and 2 were given the task of supplying information on landscape values that would be used to inform citizens' prioritization of polygons. For this, we administered the landscape valuation sheet to them, asking to fill out one sheet per polygon (Table 4). This sheet included qualitative information on each category of landscape values as well as a numerical assessment of their relative importance as compared to the other polygons. Actors were asked to work with those categories relevant to their competency or expertise and leave the rest blank.

In February and March 2016 we gathered and synthesized the inputs from the valuation sheet. We received texts on one or more of the 6 categories from 13 actors. 11 of these actors provided polygon-specific valuations, while two of them provided general information or guidelines. We received numerical assessments (1–5 ranking) from 9 actors. Two actors teamed up to perform the exercise as they belonged to the same agency. We synthesized the texts per category and polygon, while checking the veracity of information and trying to keep the original diversity and nuances. We calculated the average values per category, polygon, and actor group (Table 5). These results were sent back to the actors together with an explanation of the methods used to synthesize the information. The ranking of polygons differed between actors in group 1 (P5>P4>P1>P3>P2) and group 2 (P3>P5>P1>P4>P2).

## 3.6. Preparing participatory exhibitions in the pilot area

In parallel, a third group of actors from the wildfire governance system was selected including those operating at the scale of the 5 polygons chosen for citizen valuation, i.e. municipality and multi-municipality (Table 2). Group 3 actors included the town councils and the local forest defence associations of the municipalities included in those polygons (Table 3). Specific meetings with mayors were held in January and April 2016 to explain the project and co-involve them in the organization of exhibitions for citizen valuation. Next a joint meeting occurred

**Table 4.** Landscape valuation sheet administered to group 1 and 2 actors to value the 5 polygons selected for pilot citizen valuation. 1 = less importance, 5 = more importance. Translated from Catalan. Source: own elaboration.

Name of contributing actor:	
Polygon number:	
Date:	
A. Biodiversity and cultural heritage	Importance:
(max. 100 words)	(1–5)
B. Current socioeconomic activities	Importance:
(max. 100 words)	(1–5)
C. Cooperative fabric	Importance:
(max. 100 words)	(1–5)
D. Elements in need of special protection during wildfire	Importance:
(max. 100 words)	(1-5)
E. Potential for social and ecological economy	Importance:
(max. 100 words)	(1-5)
F. Other values	Importance:
(max. 100 words)	(1–5)

Explanation of categories: A: Intra- and inter-specific diversity as well as environmental diversity, intricately linked to the structure and dynamics of a cultural landscape including archaeological and architectonic elements, agrosilvo-pastoral management systems and intangible heritage. It can include the state, threats, potential and protection figures of all these values. B: Settlement patterns and current economic activities including forest estates (structure, land-use types, whether they have planning schemes in force, main obstacles to enhance economic activities based on local forest resources). C: Cooperation networks between people, between estates, between forest landowners and governmental agencies or between forest landowners and local and regional economic sectors. They include associations of forest owners, forest defence associations, public-private partnerships for forest planning or commercial networks of regional products. D: Elements especially vulnerable to wildfire, including houses, certain forest uses or key economic activities at the local and regional level. E: Potential for new economic activities and employment based on an enhancive use of local resources (forests, pastures, cropland, water, farmhouses, services, socioeconomic networks) as a source of wellbeing. It also includes intervention proposals to build resilience to wildfire and the explanation of their effects on wildfire behaviour. F: Any value not mentioned in the former categories.

https://doi.org/10.1371/journal.pone.0204806.t004

	Pol	ygon 1					Poly	țon 2					Polyge	3n 3					Polygo	1 <b>4</b>				Pc	olygon	5			
	¥	B	U	٩	н	щ	¥	в	с	D	Е	F	P	в	0	D	Е	, ц	A	0	-	E	щ	A	B	U	٩	ш	щ
8	u nv	r 4	4	nv	4	nv	nv	3	3	nv	2	nv	nv	5	4	5	5	nv	nv	5	4	nv	5	U N	N		4 5	ŝ	nv
۹ 	vu v	۲ 5	ŝ	nv	nv	nv	nv	5	5	nv	nv	nv	nv	5	5	nv	nv	nv	nv	5	5	nv	nv r	u n	IV	5	5	vu nv	r nv
c-d	l nv	r nv	nv	4		4	nv	Nv	nv	1	nv	2	nv	nv	nv	3	nv	1	nv	nv	nv	4	NU	3	I	nv r	5	4	ŝ
e	vu s	۲ 5	ŝ	nv	nv	nv	nv	5	5	nv	nv	nv	nv	5	5	nv	nv	nv	nv	5	5	nv	nv r	U N	N	-co	5	vu nv	vn nv
t	f 5	3	nv	5	5	4	ŝ	3	nv	ŝ	ъ	4	5	5	nv	5	ъ	5	5	4	nv	3	ۍ.	۰ ۲	4	3 r	W 4	5	4
00	3	2	nv	nv	nv	nv	ŝ	4	nv	5	2	nv	5	4	nv	5	7	nv	5	4	nv	nv	2 r	, VI	4	3 1	N N	vu n	'n
q	1 4	nv	nv	nv	nv	nv	4	Nv	nv	nv	nv	nv	5	nv	nv	nv	nv	nv	3	nv	nv	nv	nv r	N	33	nv r	N II	vu nv	vn nv
i	i 4	nv	3	5	nv	4	4	Nv	2	4	3	4	5	nv	3	5	4	4	4	nv	3	4	, vu	4	4 I	nv r	W 4	n	4
Category average	4.0	0 3.8(	4.25	4.67	7 3.33	3 4.00	4.50	4.00	3.75	3.75	3.00	3.33	5.00	4.80	4.25	4.60	4.00	3.33	4.25	L.60 4	.25 3	.67 4	.00 4.	00 3.	75 4.	.20 4.	67 4.5	0 4.6	7 4.33
actor group 1 (a-e)	n II	7 4.67	4.67	4.00	2.50	) 4.00	nv	4.33	4.33	1.00	2.00	2.00	nv	5.00	4.67	4.00	5.00	1.00	nv	5.00 4	.67	1.00 5	.00 3.	00 n	IV 5.	.00 4.	67 5.0	0 4.5	0 5.00
actor group 2 (f-i)	) 4.0	0 2.50	3.00	5.00	5.00	4.00	4.50	3.50	2.00	4.67	3.33	4.00	5.00	4.50	3.00	5.00	3.67	4.50	4.25	4.00 3	.00	.50 3	.50 4.	50 3.	75 3.	.00 r	IV 4.0	0 5.0	0 4.00
Polygon average			4	1.01					3.	72					4.3	3					4.13						4.35		
actor group 1 (a-e)			æ	3.97					2.	73					3.9.	3					4.33						4.83		
actor group 2 (f-i)			3	3.92					3.	67					4.2	8					3.79						3.95		
Actors: a) Forest	t Prot	perty C	entre.	Catal	an Dei	partme	int of /	Agricul	ture; l	) Fede	sration	1 of Fo	rest D	Vefence	s Assoc	ciatior	us Vall	ès Ori	ental (	County	; c) Sı	apport	Grout	o for F	orest	Interv	entions	GR/	AF), Fir
Department, Cat	talan	Depar	tment	of Ho	me Af	ffairs; c	l) Nor	thern M	Metroj	politan	Emer	.gency	Regio	ın, Fire	e Depa	urtmer	ıt, Cat	alan D	eparti	nent o	f Hon	r. 1e Affa	irs; e) '	Techn	ical O	office o	f Mun	icipal '	Wildfire
Prevention, Barc	celon	a Provi	nce Aı	uthori	ty; f) (	Coordi	nation	Grou	o for N	Montse	iny De	fence;	g) De	partm	ent of	Food	and A	nimal	Scienc	e, Aut	onom	ous U	niversi	ty of B	Barcelo	ona; h)	Muse	um of	Natura
Sciences in Gran	noller	s; i) M(	ontsen	y Ethi	nology	v Muse	um in	Arbúc	ies.																				

Table 5. Results of the landscape valuation exercise (quantitative part) based on input from group 1 and 2 actors. Lower case letters a to i refer to actors that provided quantitative assessments

Categories: A) Biodiversity and cultural heritage; B) Current socioeconomic activities; C) Cooperative fabric; D) Elements in need of special protection during wildfire; E) Potential for social and ecological economy; F) Other values (see description of categories in Table 4).

https://doi.org/10.1371/journal.pone.0204806.t005

13/35





**Fig 4. Common GIS.** Displaying and overlapping the GIS layers provided by the actors supported the landscape covaluation exercise and was key to build a legitimate participatory process. Here we provide two examples. a) Forest management schemes in force, provided by the Forest Property Centre. The extent of land under planned forest management served as an indication of the potential to develop joint efforts for wildfire prevention, landscape management and enhancement of regional economic activities. Overlapping this layer with the one on settlements made clear a challenge for wildfire risk reduction: the integration of forest and urban planning, currently under disconnected agencies. b) Strategic management points planned by the Fire Department and landscape management planned by the Montseny Association of Forest Landowners. The common GIS allowed identifying areas of convergence and complementarity between public wildfire prevention criteria and private landscape management interests, revealing potential synergies between actors. Source: own elaboration with data from Institut Cartogràfic i Geològic de Catalunya, Instituto Geográfico Nacional, Forest Property Centre, Montseny Association of Forest Landowners and Catalan Fire Department. Note: "Cardadeu" is misspelled. The correct spelling is "Cardedeu".

https://doi.org/10.1371/journal.pone.0204806.g004

with the mayors, some of the councillors and the heads of the local forest defence associations, where the functioning of the exhibitions was explained in detail.

One poster per polygon was designed based on the landscape valuation synthesised from actors' input (Fig 5). The synthesis text was shortened and edited for clarity. Each poster provided information on the different categories of landscape values and their relative importance as compared to the other polygons. The general average value of the polygons was not shown in the posters. Posters also included information on the wildfire and forest management tasks performed by each actor of the wildfire governance system so that citizens could learn about them. These posters were the basis of the citizen valuation exercise. Two additional posters were designed, one presenting the project (Fig 6) and another one asking participants to locate themselves in the study region by means of stickers (Fig 7).

A sheet was designed to gather participants' valuation in the exhibitions (Table 6). The sheet included a numerical assessment that consisted in distributing 15 points among the 5 polygons according to their subjective relative importance. This gave the participants the option to prioritize polygons or to assign the same value (3) to all. In this way, those participants that would not necessarily agree on the need to prioritize polygons, based on ethical or practical reasons, could still participate. The sheet also included a qualitative assessment, i.e. the reasons for their particular distribution of points. It offered the possibility of adding any missing value, and asked the participants what they learnt from the exhibition. The sheet gathered basic personal data to monitor the composition of the voters.

## 3.7. Conducting participatory exhibitions in the pilot area

Two participatory exhibitions open to all public were held in April and May 2016 in two towns of the pilot polygons (Figs 2 and 3). The two town councils co-organized the exhibitions with the facilitators and advertised them on their webpages and through their social networks. The exhibitions were also advertised through leaflets and personal networks of the facilitators. The actors involved in the previous stages of the process (group 1 and 2) were also invited to attend. The exhibition ran one and a half days in Sant Esteve de Palautordera and half day in Montseny. As participants showed up, the facilitators guided them through the exhibition and explained the goals of the participatory process. First we showed the introductory poster (Fig 6) and then the interactive one (Fig 7). Then we showed some slides explaining the logic of wildfire propagation across contention polygons and the role of SMP, as well as why a public landscape valuation was needed to inform strategies and prioritize interventions. Next we gave them the valuation sheet (Table 6) with instructions on how to fill it, and asked them to do so after reading the polygon-specific posters (Fig 5). For this, they were given unlimited time.

After the exhibitions the data from the valuation sheets was inserted into a spreadsheet. This had a row for each participant including personal data, distribution of points, and a

# ECOLOGIA POLÍTICA DELS INCENDIS FORESTALS

Po

## polígon 1: Sant Elies

















**Fig 5. Example of the posters used in the participatory exhibitions to highlight the values found in each polygon (polygon 1 poster).** The maps showed the logic of creating a common GIS, where values and land-use conflicts and synergies among actors could be visualized. As an example, in the posters we compared the map of the Montseny Association of Forest Landowners and the Forest Property Centre (upper part) with the zoning of the Montseny Natural Park (lower part; zoning currently annulled). In the right column there was a box for each category of landscape values, with a synthesis of the values found for that category, and the average numerical assessment of their relative importance as compared to the rest of the polygons. Both the text and the numerical assessment came from the input of group 1 and 2 actors. Source: own elaboration based on input from the participatory process. The maps within the poster contain data from Institut Cartogràfic i Geològic de Catalunya, Instituto Geográfico Nacional, Forest Property Centre, Montseny Association of Forest Landowners, Montseny Natural Park and Catalan Fire Department.

https://doi.org/10.1371/journal.pone.0204806.g005

summary of the qualitative information provided by him/her (reasons for the distribution of points, additional values and things learnt). Basic descriptors of the two sets of voters were created such as gender composition and place of origin. The points distributed by participants were added up per polygon. In the first town, 56 participants filled the valuation sheet, including the facilitators (4). One of the participants distributed a number of points other than 15 and another one chose not to provide a numerical assessment. Thus, 54 valid votes were collected. In the second town, 14 participants filled the valuation sheet. Overall, 68 persons voted, distributing a total of 1020 points. As shown in Fig 8, polygon #5 turned out to be the most valued one with 264 votes, followed by #3 (216.5), #1 (204.5), #4 (170) and #2 (165). The two participatory sessions yielded similar priorities, as there was disagreement only in the fourth and fifth positions (Fig 8).

### 3.8. Designing a wildfire strategy sensitive to social values about landscape

The results of the citizen valuation were then integrated in the design of a wildfire strategy sensitive to social values and priorities about landscape (Fig 3). Even if the polygons' ratings were very close to each other, some priorities were evident. Sub-polygons were drawn in each polygon to better understand the spread patterns and the potential area burnt according to wildfire behaviour and strategy (Fig 9). The worst case scenario was used, i.e. a wildfire starting in subpolygon 1.1 and propagating under west winds with all five polygons available to burn. The fire spread was drawn by distinguishing between front, flank and back propagation.

The suggested strategy consisted in working the back-left flank of the wildfire to avoid the propagation to sub-polygons 1.2 and 1.3, thus minimizing losses to the third most valued polygon. The opportunity to work on the front was instead considered to lie in the wildfire's downhill move in sub-polygons 5.1 and 5.2. If the Fire Department would succeed in these two tasks, the wildfire would be contained in the area 1.1 + 5.1 + 5.2 (Fig 9B, grey area). This would save more than half of polygon 5 and all polygon 3, the two most valued by participants. If instead the opportunity in the front would be lost, the wildfire would propagate to other sub-polygons of 5 and to polygon 4 (Fig 9B, blue or green areas depending on whether the firefighters could work on the flanks or not). It is important to underline that in a high intensity wildfire, sub-polygons 1.1, 5.1 and 5.2 are considered to be beyond the extinction capacity, hence not workable.

To make this strategy possible, the implementation of two SMP was suggested (Fig 9C). The first one is located in the valley bottom at the centre of polygon 5 and falls within the municipalities of Sant Esteve de Palautordera, Fogars de Montclús and Sant Pere de Vilamajor. It would facilitate firefighting operations to avoid that the wildfire moves uphill to the opposite slope of the valley. For this, it was considered necessary to achieve a discontinuous vegetation structure preventing crown spread and thus limiting wildfire intensity so that the firefighters can work. The second SMP is located before the crest line dividing polygon 1 and 5, and falls within the municipalities of Sant Esteve de Palautordera and Sant Pere de Vilamajor. Lower fuel load in this SMP would slow down the wildfire's uphill move and thus reduce spotting distance to polygon 5.

# ECOLOGIA POLÍTICA DELS INCENDIS FORESTALS

## panell general: cooperació ciutadana en el territori



### DESCRIPCIÓ DEL PROCÉS

#### PER QUÈ AQUEST PROCÉS PARTICIPATIU

a Universitat Humbolid de Berlin, Lambientéleg i investigador lago Ofere coordina un projecte titulat " <b>Lecelogia politica dels incen-</b> <b>Serestata</b> " que estudia els processos de agrementatge social en materia de provio di nucendis forestats. El cas d'estud es Catalun- i més concretament, els massicos del Montseny, el Corredori el Montnegre.	Hem rebut aportacions molt diverses i totes rellevants de: Associ Federació d'ADF del Vallès Oriental, Bombers de la Generalitat, Diputació de Barcelona, Serveis Territorials del Departament Centre de Recerca Ecológica i d'Aplicacions Forestals, Departa
na part essencial d'aquest projecte és aquest procés de participació, codissenyat amb els propis agents del territori i les administra- uns competents, per tal de millorar la coordinació i les polítiques de prevenció i extinció d'incendis forestals. Alhora, el procés vol reco-	Montseny La Gabella d'Arbúcies i Museu de Ciències Naturals o
elements de valoració ciutadana del paisatge per a integrar-los en el disseny de les estratégies de prevenció i extinció.	De tots ells hem rebut textos sobre una o més categories de valor fiques per poligons, però també hem rebut valoracions de caràcte
procés vol implicar els diferents actors amb competències i interessos en la matèria, així com la ciutadania, <b>en la co-responsabilitza-</b> i l la cura compartida dels boscos per fer front als grans incendis forestals.	COM N'HEM FET LA SÍNTESI?
IB QUI HO HEM FET?	Primer hem integrat tots els textos per poligons i categories, tot si dessin reflectits
uest procés busca construir la base i l'impuls per a una presa de consciéncia ciutadana en relació al potencial de cohesió política i te- orial que hi ha en les propostes de gestió i custódia del territori.	Després hem introduit les valoracions numériques en una fulla de
r això, ha estat fonamental determinar, contactar i treballar amb els agents ciutadans i de l'administració que estaven implicats en uesta custódia. Associacions de propietaris forestats i agrupacions de Defensa Forestal, d'una banda, i administracions compe- nte de l'atra Divinsició de Barcolana Generalità de Gatalumos adricom centres de recerca i acumacions de la societat ciul re-	jana resultant com a guia i recolzament de la valoració ciutadana. I polígon en relació als altres polígons per cada categoria de valors
vants en la qüestió de la conservació dels boscos i el patrimoni natural i cultural.	COM FUNCIONA LA JORNADA DE PARTICIPACIÓ?
nb ells hem codissenyat les eines i hem consensuat la metodologia del procés de participació ciudadana, que hem acabat de validar ho els ajuntaments de l'ambit més reduit que hem escolit per a la prova pilot de la dinàmica participativa (Ajuntaments de Sant teve de Palautordera, Montseny, Fogars de Montciús I Sant Pere de Vilamajor)	La jornada de participació s' ha previst com el dispositiu per inclou i criteri.

#### COM HO HEM FET?

F

- En un primer momeri, i del a mà del grup d'especialistes en incendis forestais GRAF (Grup de Recotament d'Actuacions Forestata) de cos de Bornbers de la Generalitat de Catalunya, es van determina en l'ambit de treball, els poligons de contenció d'incendis, segon el comportament d'un incendi potencial i les possibilitats de les estratègies d'extinció.
- que els agents implicats en el procés han aportat.
- Es tractava de fer una valoració conjunta dels diferents poligons. I per això cada poligon disposa d'una fitxa pròpia redactada pel conjunt d'agents i administracions competents segons les següents categories:
- B. Activitat socioeconòmica actual
- D. Elements de protecció especial en cas d'incendi forestal
   E. Potencial d'economia social i ecològica

#### COM HEM FET ELS CONTINGUTS DELS PANELLS?

Allarg de mesos hem and conductint i tenirt enuices de traball amb tots été agents que des de cada un dels seus refs i amb d'accès insponsabilitat une cura del tenirror forsatul. La fraitat d'aquesta avais de traballin estat, d'una borda, la de companit e les bases cartopràfiques de cada un de nosatires per fer-ne una de comuna. I d'atira banda, la de recalta conjuntament els continguts per ta de recult a complementantetal d'unestata de valenciares respectes del serroria.

#### QUINES APORTACIONS HEM REBUT?

Hen rebut aportacions moit diverses i totes relevants de Associació de Propietaris del Montseny. Centre de la Propietat Forestal, Federació d'ADF del Valles Oriental, Bombers de la Generalitat, Oficina Tecnica de Prevenció Municipal d'Incendis Forestala de la Biojutació de Barceinos, Serveis Territoriolas del Departament d'Agricultura, Coordinadro par e la Salvaguarda del Montseny, Centre de Recerca Ecológica i d'Aplicacions Forestals, Departament de Ciència Animal (dels Aliments (UAB), Museu Etnológic del Montseny La Gabella d'Arbuicis Museu de Ciències Naturals de Granollers :

De tots ells hem rebut textos sobre una o més categories de valors. En la major part dels casos aquests textos són valoracions especifiques per polígons, però també hem rebut valoracions de caràcter general.

Primer hem integrat tots els textos per poligons i categories, tot sintetitzant la informació aportada i mirant que tots els matisos hi quedessin reflectits.

Després hem introduit les visionacions numériques en una faita de adacta per poligen categoria i acto. D'aquesta maniera, hem poput ferter mitjanse de visions per categoria per poligoni consolutivas ente hem pout natera en cada paneli categoria al valancia mitjans resultant com a guía i recolament del a visionació cutadana. La valoració numérica volse una indicació de la importincia d'aquell poligio en relació a alimes poligone per cada categoria de vales. Exectal va de la fu poli meter esta de la fuel en terratoria.

a jornada de participació s' ha previst com el dispositiu per incloure també en el procés a la ciutadania i recollir les seves valoracions criteri

Es tracta d'una exposició de 6 panells informatius a través de la qual la ciutadania té accés finalment al resultat d'aquest trebail en xanxa que hem estat realitzant durant mesos i pot indicar, amb la serva valoració dels poligons, quina ha de ser (per compartida i consensuada) la millor estratégia de prevenció i extinció d'incendis.

Aquests panells informatus han de servir també perquè totes los persones que participin puguin conèxer aspectes relacionats amb els valors dels sexu paísatges, et comportament dels incendis que s'esperen i les mesures de prevendo executades o proposades També es busca facilitar la familiaritzacio amb les tasques que porten a terme els diferents actors o institucions experts en la matería.

Després d'haver llegit és platfors amb és objectius del projecte la informació sobre els polgons. Ni ha la possibilitat d'omplir la butlitet a de valoració dels polgons, distribuint un total de las jounts entre toto ella Podeu distribuir-los com vulgueu però els heu de gistar tots! Això us dona moltes opcions: des de prioritzar-ne alguns fins a fer-los empatar.

#### SPRES GOE? COM SEGUIM?

Un cop consguida ia valoració que la citutadamia ha fet de cada un d'aquests polgons: es pot passar a fer una prioritzacio dels matinosos rientarna a Bomberso els susultas prevaio determinin i proposi quines han de set les sistatejos els egois lo prevento sogones del difrento secenaris d'incendis que podem esperai. La proposta de Bombers serà posteriorment consensuada amb els diferents actors implicats en el proceso cooperatu. Un con consensuada la proposta, els ponsa en fer un retorn dels resultats a totes les persones que hagin deixa indicat en les seves butiletes que volen seguir essent informades.

#### es i consensuades abans que aquest es produeixi, i athora poder dissenyar mesures de prevenció i coneficients i coordinades.

Per això creiem que amb aquest procés estem co-decidint entre tots i totes, l'ordre d'actuacions, tant en el moment de l'incendi, com en el moment de gestionar les accions per a la seva prevenció.

#### معام المعالية المطلبة

Fig 6. Poster used in the participatory exhibitions, containing general information on the project. The poster included the study region, goals, actors, method of landscape co-valuation in wildfire contention polygons, public participation procedure, and subsequent steps. Source:

own elaboration based on input from the participatory process. The maps within the posters contain data from Institut Cartogràfic i Geològic de Catalunya, Instituto Geográfico Nacional, and Catalan Fire Department.

https://doi.org/10.1371/journal.pone.0204806.g006

## 3.9. Discussing the strategy and the next steps with actors and participants

In May 2016 a joint meeting was held with group 1, 2 and 3 actors, where the participants of the exhibitions were also invited. First, we presented a summary of the participatory process, the composition of the two sets of voters and the quantitative and qualitative results of the citizen valuation. Next we presented and discussed the proposal of value-sensitive wildfire strategy and related SMP (Fig 9), which was agreed upon in the same meeting. Finally we identified some applications of the pilot project and discussed proposals for next steps (Table 7). Some of these applications stressed the interest of further developing the multi-actor GIS to coordinate interventions and adapt wildfire strategies to changing landscape structure and fuel conditions, while others emphasized that the project's materials and network could be used as vehicles for knowledge dissemination on wildfire prevention and civil protection.

The actual implementation of the agreed upon SMP was suggested, but a consensus emerged that the mayors of the 4 municipalities involved in the participatory exhibitions should take the lead in the continuation of the project. Several actions to do so were suggested, such as inviting the rest of the mayors of the Montseny mountain ranges to join in a second phase (Table 7). In the meeting, the project was in general considered a first step towards a Fire Department that is able to protect landscape values as well as people and properties, and it was deemed interesting to present the experience to the boards of some of the involved institutions. It was considered crucial to involve experts and policy-makers in land and urban planning to integrate wildfire risk into planning schemes at different scales.

Finally, a meeting was held in June 2016 with the mayors, councillors and heads of local forest defence associations to discuss the intervention proposals relevant at the municipal and multi-municipal scales (Table 8). It was organized by the mayors following one of the agreements of the previous joint meeting (Table 7). It was considered necessary to share the cartographic information on SMP and the contact details of participants for coordination purposes, as well as to set up a virtual platform with the multi-actor GIS accessible to the town councils' technical services. The organization of dissemination activities on wildfire prevention for vulnerable residential areas was suggested, and the mayors committed to requesting a meeting with the Natural Parks division of the Barcelona Province Authority to budget the implementation of the priority SMP. To ensure the long-term economic feasibility of maintaining appropriate forest structures in the SMP, it was considered that the firewood could be used to supply biomass boilers already existing in municipal equipment facilities for heating, while creating links with the biomass sector in the region.

# 4. The participatory process: Democratic and transformative wildfire planning?

### 4.1. Position and legitimacy. Participating in what?

GRAF was both an agenda setter and an invited actor in the participatory process. This situation stemmed from the first author's long-term collaboration with GRAF in wildfire research and management projects. Being sympathetic to GRAF's vision on wildfires, the first author had to perform a "neutral" role of facilitator among actors while "siding" with one of them. This contradictory position raised important questions about potential biases in the design and implementation of the participatory process. We controlled for these biases by i)

# ECOLOGIA POLÍTICA DELS INCENDIS FORESTALS

## cartografia de la participació ciutadana



#### FES-TE VISIBLE EN EL TERRITORI !

Amb uns breus minuts d'interacció teva amb aquest panell, podrem fer visible la vostra presència i moviment en el territori que estem analitzant en aquestes jornades de participació.

Aquesta cartografia vol deixar traça de l'us real del territori per part dels seus habitants i d'aquelles persones que sovint el visiten.

Indican's (amb un gomet groc) on vius o en quin lloc t' estàs quan vens al Montseny.

Indican's també (amb un gomet de color verd) els llocs dels que gaudeixis més d'aquest territori i també els que més hi vas.

gràcies per la teva col·laboració!

PLOS ONE

Fig 7. Interactive poster where the participants of the exhibitions were asked to attach a yellow sticker in their place of residence and green stickers in those places that they used (for shopping, leisure, etc.). On the right, we have superimposed pictures of this poster after the exhibition of 23–24 April (above) and 4 May (below). Source: own elaboration based on input from the participatory process. The maps within the posters contain data from Institut Cartogràfic i Geològic de Catalunya, Instituto Geográfico Nacional, Montseny Natural Park (zoning currently annulled) and Catalan Fire Department.

https://doi.org/10.1371/journal.pone.0204806.g007

characterizing the challenge, ii) implementing the co-design principle, and iii) reflecting on the actual reach of actors' and citizens' participation.

The challenge was to democratize GRAF's decision power over the region's landscape by means of a participatory method that was both operational, i.e. translatable to concrete wildfire strategies, and legitimate, i.e. integrating the values and management tasks of other actors in a way that was meaningful for them. For this, we implemented the co-design principle, based on reaching agreements with actors on what was to be done and how during the entire process, from the initial individual meetings to the final decisions about how to continue after the pilot phase. This required flexibility to adapt the process' development to actors' changing engagement in it. The landscape valuation method was for instance co-designed with actors and tested in joint meetings, including what to value and how, contradictions between values, whether prioritization of polygons was possible at all and which criteria should be used for this. The process likewise accommodated actors' attempts to turn it into something useful for their interests, something that implied building alliances with them. For instance, the forest owners saw the process as a potential tool to enhance forest management, disseminate their role as land stewards, better allocate public funding for forest management and simplify administrative procedures. For the Forest Property Centre, the results of the participatory process could be used to prioritize public funding and to negotiate with forest owners on special management conditions required to implement the strategic management points (SMP). The Barcelona Territorial Service of the Department of Agriculture was interested in the process as a way to improve inter-agency coordination. The process fit in the general interest of both Natural Parks to improve governance through more democratic decision-making bodies. For the Technical Office of Municipal Wildfire Prevention of the Barcelona Province Authority, the process could be an opportunity to better establish which interventions in private estates are financed with public money and which ones with private money.

Table 6. Sheet given to citizens attending the participatory exhibitions. Citizens were asked to fill in this sheet after reading the posters on landscape values prepared with the input from actors. Translated from Catalan. Source: own elaboration.

Personal data	
Name (optional):	
E-mail (optional):	
Would you like us to inform you about the next steps and the results of the project?	
Place of residence:	
Do you belong to an active organization in the study area? Which one?	
Valuation	
Please distribute 15 points among the 5 polygons according to their importance for you. You and no more than 15 points, but you can distribute them as you wish.	must use all 15 points
P1 P2 P3 P4 P4	P5
Please explain us the reasons of your valuation:	
What did you learn from reading the posters in this exhibition?	
Would you like to add a value from the polygons that was not mentioned in the posters?	
The Political Ecology of Wildfires	
Place:	Date:
https://doi.org/10.1371/journal.pone.0204806.t006	





https://doi.org/10.1371/journal.pone.0204806.g008

A clear example of how actors shaped the process according to their own interests–legitimizing it along the way–is the organization of the second participatory exhibition, which was not planned by the facilitators but proposed by a councillor of Montseny municipality. When he attended the first exhibition–occurring in a town of the plain–he invited us to bring it to Montseny–a mountain municipality–and took charge of the dissemination. The board meeting of the local forest defence association was scheduled immediately after the exhibition and in the same venue so that its members could participate and vote. Not surprisingly, this participatory session turned out to be mostly composed of mountain settlers and members of the forest defence association. As the poster locating participants' residence was filled with yellow stickers in the mountain areas (Fig 7), the councillor told us "You see? Now it is balancing". Even if the councillor's move could be read as an attempt to lobby the process, it turned out to balance it in terms of participants' place of residence and sociological profile (Section 4.2).

The creation of a common GIS made the co-design principle visible and believable for the actors. When forest owners said that GRAF's wildfire contention polygons were not meaningful for them, the facilitators decided to ask all actors those GIS layers representative of their wildfire and forest management tasks. A GIS was subsequently built with layers from almost all group 1 actors including private and public properties with forest planning schemes, wild-fire and landscape management interventions by forest owners, forest tracks and water infrastructures maintained by public agencies, and zoning of natural protected areas (Fig 4). These layers were overlapped with GRAF's (Figs 2 and 4B). In the first joint meeting with the actors, the GIS layers were displayed, commented by each actor and discussed to highlight common interests and potential synergies. The actors, including GRAF, could explain their needs and expectations in a transparent way. This served to recognize the existing diverse perspectives on wildfire and forest management, while stressing-and legitimizing-the need of synthesizing



**Fig 9. Summary of the wildfire strategy designed by GRAF and discussed with the actors.** a) The strategy was prepared for a wildfire starting in polygon 1 (red star) and driven by west winds. Red arrow: front spread (high intensity); yellow arrow: flank spread (medium intensity); green arrow: back spread (low intensity); blue lines: wildfire contention sub-polygons, used to estimate the potential area burnt. b) Potential area burnt according to the success or failure of the strategy, as well as fuel conditions. Grey, blue and green areas correspond to a wildfire starting in sub-polygon 11 (discussed with the actors). The orange one corresponds to another starting point, which could eventually be discussed in future exercises. c) Strategic Management Points (SMP) that need to be developed to reach wildfire friendly landscape structures and make possible the strategy under discussion. Developing the SMP in polygon 5 would provide an opportunity for the Fire Department to contain the wildfire in the valley bottom. If this would work the burnt area would resemble the grey shape in Fig 9B. Instead, if this SMP is not developed, the burnt area would resemble the grey shape in Fig 9B. Instead, if this SMP is not develop the SMP between polygon 1 and polygon 5 to reduce spotting distance and facilitate wildfire confinement in the grey shape. Source: own elaboration with data from Catalan Fire Department.

https://doi.org/10.1371/journal.pone.0204806.g009

them at the scale of wildfire contention polygons to inform the strategies of the Fire Department.

The above leads us to reflect on the actual participation of actors and citizens in the process. They co-valued the landscape and prioritized polygons to inform the Fire Department's strategy, which included choosing among a set of initially defined SMP. Yet they did not participate in the production of knowledge about wildfires and how they should be fought. The role of wildfire experts was assigned to the Fire Department and, in particular, to GRAF, whose wildfire contention polygons and SMP framed the valuation exercise from the beginning. Even if the polygons and the need to prioritize according to social landscape values were intensely discussed with actors in the first joint meeting, GRAF's technical knowledge and overall framing were not contested, reproducing its hegemonic position in the Catalan wildfire governance system [1]. However, while the participatory process did not question GRAF's knowledge, it opened expert wildfire management to society by incorporating landscape values into the Fire Department's strategies.

## Table 7. Project applications and proposals to continue networking as discussed in the joint meeting with all actors in May 2016. Source: own elaboration.

Use the common GIS developed during the project to monitor management interventions and to adapt the Fire Department's wildfire strategies to the landscape's changing state. Create a virtual platform and give access to all actors so that current and future intervention proposals can be jointly operationalized.

Integrate selected strategic management points (SMP) in landowners' forest planning schemes.

Use the project's planning network as a vehicle of knowledge exchange in civil protection, for instance by disseminating information on self-protection to neighbours of vulnerable residential areas.

The Fire Department can protect landscape values as well as people and properties.

Implement the selected SMP with the collaboration of the different actors.

Involve the Catalan Department of Territory and Sustainability, responsible for urban planning, in subsequent steps of the project. Incorporate existing planning schemes in the common GIS to check obstacles and opportunities for the integration of wildfire risk into urban planning.

Organize a specific meeting with the mayors of the municipalities included in the 5 pilot polygons to discuss the continuation of the project at the local scale.

Suggest to those mayors that they present a more concrete version of the project proposals to the board of the Natural Park and Biosphere Reserve of Montseny.

Present the project experience in the board of the Forest Property Centre, composed by representatives of different Departments of the Catalan government.

Expand the project by replicating the pilot valuation to all Montseny mountains. The mayors of the pilot municipalities may invite the rest of the mayors to join such a project.

Disseminate knowledge by circulating the posters used in the participatory exhibitions around different places in combination with dissemination efforts by the Fire Department and the Federation of Forest Defence Associations.

Distribute a copy of the posters used in the participatory exhibitions among the project's network of actors and citizens.

https://doi.org/10.1371/journal.pone.0204806.t007

 Table 8. Some agreements and proposals discussed with the mayors and other group 3 actors in the final meeting of the project.

 Source: own elaboration.

The mayors request a meeting with the Natural Parks division of the Barcelona Province Authority to ask financial and technical support to design and implement the priority strategic management points (SMP).

GRAF transfers the GIS with the SMP to the town council's technical services.

ICR makes a budget to set up the virtual platform with the common GIS, with access to all participants.

The firewood resulting from the implementation of the priority SMP can be used to supply existing biomass boilers in municipal facilities. Income from firewood can be invested in the maintenance of the SMP. Collaboration with the Montseny Association of Forest Landowners and a nearby biomass cooperative could invigorate the region's biomass sector.

Several dissemination activities can be co-organized by the town councils and the Fire Department on prevention and self-protection, for residents in vulnerable developments.

IRI THESys disseminates a press release on the project results.

https://doi.org/10.1371/journal.pone.0204806.t008

Indeed, the participation of actors and citizens had large implications. Opening GRAF's decision-making power to wildfire governance actors, landscape inhabitants and landscape users, was not always easy. Some actors agreed to help the Fire Department to make more informed decisions during wildfires, but still stressed that it is this agency which is responsible for these decisions. In the participatory exhibitions, citizens spent a long time reading the posters and thinking how to distribute the points in the valuation sheet. This suggests that they were not only aware of the complexity of the valuation but also of the implications of their participation for actual decision making. However, making an informed decision required time and also the capacity to process multidimensional information. Some participants, for instance, complained about not having had enough time to read the posters. It is thus likely that in many cases the valuation was done after assimilating only a share of the information provided. Moreover, the ways in which the participatory landscape valuation would be translated into a concrete wildfire strategy were not well communicated because at the time we did not know how such a translation would be done.

In the end, the aggregated numerical assessment (Fig 8) turned out to be the only criterion used to design the wildfire strategy. Neither the qualitative valuation and the additional values reported by the participants nor the values and management proposals provided by the actors in the posters had a direct translation into the strategy. Moreover, even if polygon #5 turned out to be the most valued one, the Fire Department committed to save only half of it, arguing that under a high intensity wildfire the other half was not workable for the firefighters (Fig 9). This suggests that the capacity of participants to decide how a wildfire affects their landscape was in actual fact limited. The feedback meeting explaining how the valuation had been translated into a strategy was thus important to round off the exercise. Still, future exercises should provide clearer information on how the valuation will be translated into concrete strategies, including the technical limits of extinction.

### 4.2. Who participated in the process and who did not?

All the actors that we considered part of the wildfire governance system (Table 2) participated in the process except the Federation of Forest Defence Associations of the Maresme County, which turned out to be of less importance as the pilot polygons were located in the Vallès Oriental County. However, some institutions with competencies in wildfire governance were not included in our sketch of the wildfire governance system. One of them was the Civil Protection Authority of the Department of Home Affairs, which is in charge of drafting wildfire emergency plans that coordinate the different agencies involved when a wildfire takes place (INFO-CAT plan). In our study region, some of this Authority's competencies are de facto transferred to the Technical Office of Municipal Wildfire Prevention of the Barcelona Province Authority, which supports municipalities in prevention and emergency planning. As this agency was included in our process, the non-participation of the Civil Protection Authority did not necessarily entail that its role in wildfire governance was neglected. Another institution that was not included was the Service of Forest Rangers of the Department of Agriculture, which function (scientific investigation of wildfire causes) we considered less relevant for the purposes of our process. The Urban and Land Planning Authority of the Department of Territory and Sustainability was also not included, despite the fact that it could become a key actor promoting the integration of SMP into legally binding urban and land planning schemes. The need to invite this agency was identified by the participants of the process, but the facilitators were not able to achieve this. As these three agencies were not present in the participatory process, the outcomes of our project could eventually be less relevant for them.

Beyond the wildfire governance system, wildfire risk is influenced by and influences a wide array of actors. Representatives of infrastructures crossing our study region (AP7 highway, international high-speed train, conventional train and electric lines) as well as industrial and touristic activities occurring in the region were not included in our process due to time and resource constraints. Similarly, the representation of potentially affected interests outside our study region due to a cut in infrastructures after a wildfire event inside our study region was not deemed feasible, but should be considered in future processes.

Actors participated to varying degrees ranging from only attending the meetings to actively contributing to the landscape co-valuation. In the test phase we got input from 6 out of the 10 invited actors from group 1. In the pilot co-valuation we got input from 13 out of 15 actors from group 1 and 2 (excluding those with a focus on Montnegre-Corredor). As actors had overlapping interests and expertise, contributing actors partially compensated for the missing information from non-contributing actors. Redundancy across actors was thus important to avoid neglecting certain landscape values. Still, redundancy was not a guarantee that all potentially relevant information was considered. The Natural Park of Montseny, for instance, did not provide any input on landscape values. This was partially compensated by the biodiversity assessment provided by the Museum of Natural Sciences in Granollers and the information on cultural heritage provided by the Montseny Ethnology Museum in Arbúcies, showing the usefulness of inviting a set of additional actors (group 2). However, the distribution of an endemic amphibian species of high conservation value (Montseny brook newt, *Calotriton arnoldi*) was for example missing because the Granollers expert on amphibians was not in the internal meeting where the assessment was drafted.

The participatory exhibitions attracted primarily residents of the municipalities were they were held (29% of participants of exhibition 1, E1, and 57% of participants of exhibition 2, E2). Even if participants also came from the other municipalities included in the pilot polygons (13% of E1 and 7% of E2), the two municipalities that did not hold exhibitions (Sant Pere de Vilamajor and Fogars de Montclús) were underrepresented. Exhibitions also attracted participants from other municipalities in the Montseny region, outside the pilot polygons (38% of E1 and 29% of E2) and from Barcelona city (13% of E1), entailing that the local landscape valuation was partly done by extra-locals. An important share of participants was formed by mayors and councillors, mostly from the municipalities in the pilot polygons (16% of E1 and 29% of E2).

The two sets of voters were qualitatively different. The first one was mostly composed of residents in the valley bottom, was gender balanced, had a significant share of anti-capitalists and ecologists (13%), and a relatively low share of participants linked to local forest defence associations (11%). In contrast, the second sample was mostly composed of mountain residents including forest owners, was male dominated (86% of the participants), and had a

relatively high share of participants linked to forest defence associations (50%). The sociological composition of both sets of voters was conditioned by the dissemination. The first one was mostly done by I.O., who attracted his activist network, and by two co-organizing town councils through their social networks, excluding people with no access to these communication tools. The second one was instead done by the hosting town council through letters to neighbours and the board of the local forest defence association. In both cases dissemination did not reach all residents in the four municipalities. Despite differences in sociological composition, the landscape valuation in the two exhibitions was surprisingly similar (Fig 8).

## 4.3. Valuing landscape and learning

A number of questions emerge around what was valued, by whom, and how. Regarding the first point, the content of the valuation was contingent upon the actors included and their availability to provide input. As such, it was necessarily non-exhaustive and relevant values potentially influencing public priorities could have been left out, exemplified by the endemic amphibian species. Regarding who defined the value categories, categories such as 'Biodiversity and cultural heritage' or 'Current socioeconomic activities' were created based on the actors' input during the test, while others such as 'Cooperative fabric' and 'Potential for social and ecological economy' were created by the facilitators to give the process a transformative direction (Table 4). The citizens participating in the two exhibitions had no influence over the value categorisation, and their function was limited to the prioritization of polygons based on the qualitative and quantitative information provided by the aforementioned actors.

Regarding how landscape was valued, heterogeneous methods and criteria were used by the participants in the process. While some actors used numbers to assess the relative importance of polygons during the co-valuation, others provided only texts. The ranking of polygons differed between group 1 and group 2 actors (Table 5), although we did not check whether these differences were statistically significant. Similarly, the citizen valuation changed the prioritization of polygons done by actors, but not dramatically (actors: P5>P3>P4>P1>P2; public: P5>P3>P1>P4>P2). In the participatory exhibitions, citizens reported to use a diversity of criteria to distribute the points. These included biodiversity, aesthetics, emotional, private property, wildfire propagation, presence of houses and developments, potential for social and ecological economy, cultural heritage, as well as unwillingness to prioritize. Additional values reported by citizens included forest type and age, a public rural school, the regenerative potential of the land, intervention proposals such as silvopasture or biomass production for energy, threats due to development projects, immaterial cultural heritage on fire, an ancient tree, and the endemic amphibian that was left out in the co-valuation. Ranking polygons based on their aggregate score thus synthesized heterogeneous and potentially contradicting values into a collective prioritization to inform the wildfire strategy. However, the use of a numeric synthesis of collective priorities may have detracted from more qualitative discussions of values and policy goals, even if qualitative information was prominent in the posters and deliberation on policy proposals occurred in the meetings.

In the valuation sheet, citizens participating in the exhibitions reported to have acquired more and better knowledge on a wide diversity of landscape values as a result of being involved in the exercise, sometimes increasing their awareness of the environmental importance of the areas where they live or spend leisure time. Several participants stressed the interest of learning to value the landscape from the perspectives of different actors. They reported to have learnt the existing wildfire types and the logic of wildfire spread and contention polygons. They also learnt how the Fire Department develops the wildfire strategies, as well as the relevance of the wildfire prevention and forest management tasks developed by those actors linked to forest owners (local forest defence associations, the Montseny Association of Forest Landowners and the Forest Property Centre). Participants reported to have learnt wildfire prevention and land management proposals too, including the recovery of pastures and cropland, the importance of forest and territorial planning, and the need of a multi-actor planning of wildfire prevention policies that integrates diverse criteria and social interests about landscape. Some participants learnt that protection from wildfires can enhance economic activities and that the ways in which society lives with nature shape wildfire risk. All these were learnings that we wanted to enhance with the participatory exhibitions. We also wanted to disseminate the idea that wildfire is a part of the ecosystem that can be managed for positive ends, and that wildfire prevention entails a re-organization of human-environmental interactions, but according to the feedback gathered in the valuation sheet these insights did not reach the participants.

## 4.4. Implementing the outcomes of a pilot process?

Ideally, to implement the wildfire planning proposals stemming from our process, the call for participation should have reached 100% of the electoral roll by official means, and the decisions made should have been approved by the town councils' plenary. Still, given the efforts made and the high wildfire risk in the region demanding urgent interventions, trying to implement the chosen SMP regardless was tempting. But when interventions had to be agreed upon, some difficulties emerged. The Technical Office of Municipal Wildfire Prevention of the Barcelona Province Authority refused to fund coordinated measures. Underlying this decision was the conflict between the office and the Montseny Association of Forest Landowners, which unlike the one in Montnegre-Corredor, does not collaborate in the office's forest planning for wildfire prevention (Table 3). The Montseny forest owners instead stressed the need to implement the SMP, which they considered legitimate enough, to reduce wildfire risk. Other actors warned that this would require building further legitimacy, with mayors taking the lead and conducting a more systematic public participation program. An agreement seemed to emerge between the two positions: the SMP could be implemented in parallel to new efforts to increase the democratic content of the participatory process. After the last meeting, the facilitators retired from their leading roles and observed the endogenous dynamics generated by the process. After 2 years, the mayors have not carried on the work necessary to implement the wildfire planning proposals, but the forest owners have held some preparatory meetings to do so.

## 4.5. Conflict, consensus and the re-framing of the wildfire issue

In our meetings, it was evident that the invited actors' framing of the wildfire issue went beyond the traditional suppression-based approach and instead thought of it as a matter of forest and landscape management. The participants of the exhibitions likewise learned that wildfire prevention is linked to a number of land management proposals such as silvopasture. However, during the process there was not enough reflection on which politic-economic policies are necessary to make such management proposals possible in a largely urban and industrial context where agro-silvo-pastoral activities are not profitable anymore. The Coordination Group for Montseny Defence stressed in the debates that recovering rural activities in the mountain required facilitating young people access to land and debating about the concept and laws of land property. This was an opportunity to reframe the wildfire issue in politically transformative ways that was however not given sufficient attention by the facilitators. Indeed, as the networking with the practitioners of alternative forest and land management practices failed, there was not a critical antagonist mass to reframe the wildfire issue as a problem of political-economic and land-use model that needs to change if we are to build a resilient landscape. Instead, the debates were mostly chaired around the need to inform the strategies of the Fire Department through landscape co-valuation, and how to operationalize it. For this, consensus among actors was always reached. Consensus was however not only due to our facilitation, but built on previous collaborative wildfire prevention projects between some of them. Conflictive issues within the wildfire governance system, such as the integration of local forest defence associations in the Fire Department's suppression operations, surfaced during debates on valuation but never derailed the consensus. Existing conflicts between actors did not play out strongly in the co-valuation exercise or in the citizen valuation–but interestingly did so when interventions had to be agreed upon.

The above suggests that in future processes a better identification and management of potential conflicts and disagreements among participants may be advisable to strengthen the transformative power of the network, both to articulate potentially re-framing visions and to ensure that the process' outcomes can be actually implemented on the ground.

## 5. Discussion

Various papers have emphasized the potential of participatory planning networks to build wildfire resilient landscapes in the face of current and future impacts of suppression and land-use and climatic changes [3–7]. Studies have stressed the need to understand resilience as a process of adaptive governance mediated by institutions at multiple scales, as it opens social opportunities to learn from and adapt to wildfire [11,46]. In turn, building resilient landscapes has been shown to require fundamental transformations in social-ecological relationships [47,48], including fossil-fuel based food and energy systems and the ensuing high risk landscapes [1]. For this, authors have stressed the need to create sites of "healthy contestation" whereby the political economy of land use is opened to debate and antagonism is channelled into transformative avenues [20,49,50]. The creation of "political communities" able to deal with conflicting values while shaping their own fire regimes has been likewise suggested as a way forward [8,9].

Our experience shows how a planning network incorporating landscape values and local knowledge into operational wildfire strategies can be designed and implemented in a fireprone region. By planning and developing links between academic and non-academic actors across scales, the pilot project also illustrates the potential of action-research [35,36] and transformative project co-design [33] for building resilience through the setup of ad-hoc political communities, as well as some of the main challenges ahead.

Our method complements existing approaches that incorporate social values about landscape and local knowledge into operational wildfire management plans and decision support systems [16–18]. Our prioritization of polygons based on an aggregate score offers a relatively simple way to translate complex social values into concrete wildfire strategies. In turn, the spatially explicit representation of personal values developed by [18] could be used in a follow-up of our pilot process, in order to also translate participants' qualitative valuation and management proposals into more inclusive wildfire strategies. Interestingly, we did not find big disagreements on social priorities between two very different sets of citizens, something that has been also reported in similar experiences [18]. This suggests that the conflicting nature of wildfire policies reported by some studies [20,51–55] might not necessarily hold in the particular case of wildfire planning networks, where basic consensus seem to emerge around what is to be protected first. Future efforts should therefore explore whether this consensus also exists when wildfire strategies are democratized beyond the pilot (micro) scale.

Our pilot process shows that the exclusion of relevant actors, citizens and values was not only due to the practicalities of the project, but also to the extremely broad range of interests potentially affected by the large wildfires expected in the region. This fact points at a fundamental trade-off between high wildfire risk levels and democracy that should be considered in further efforts to democratize wildfire management [9]. Our experience however suggested some offsetting mechanisms, including value redundancy across invited actors, and a continuous improvement of the network's democratic content while the (imperfectly) agreed upon interventions are implemented. This in turn requires resources and incentives to continue networking and to monitor the interventions on an ongoing basis [17], something that is not always provided by the institutional arrangements and funding mechanisms of universities or public agencies [56]. Indeed, when our project ended the network hardly continued its activities, suggesting that more efforts should be devoted to secure the follow-up of pilot processes [18]. A key follow-up stage should involve the creation of the actual decision support system including the multi-actor GIS, the valuation results and the agreed strategy and interventions. Using these elements in combination with tools that facilitate stakeholder's deliberation about alternative fire management policies [9,11] could potentially enhance the network's transformative power.

Our experience reveals how the connection of different governance scales through a wildfire planning network (re-)defines agency over landscape, i.e. the power to actively shape landuses, values and wildfire impacts while building resilience. Priority interventions were defined at the local level by both local and extra-local citizens and actors. In turn, the ability of citizens to define the wildfire strategies that would transform their living environments for decades was constrained by suppression limits and by the participatory method itself. Therefore, the positive link between local empowerment and better wildfire governance reported by some studies [8,17] should be problematized according to the varying ways in which planning networks are operationalized across scales. Rather than considering local agency and bottom-up processes as the privileged realm for transformative action, our experience suggests that wildfire planning networks could benefit from a strategic mobilization of the knowledge available at different scales [57]. Indeed, the combination of top-down and bottom-up valuation mechanisms gave our process structure and direction while leaving space for autonomous development and learning, as it has been found in other collaborative planning contexts [58]. These insights suggest that participatory and transformative planning networks may offer the kind of adaptive governance structures and processes needed to effectively deal with wildfire risk in the face of climate and land-use changes [11,46].

The pilot project also illustrates how participatory planning networks might change the relationship between expert knowledge and agency over landscape as the wildfire strategies are democratized. As explained in our critical reflection on the participatory dynamics, GRAF knowledge was not negotiated and framed the whole process. But our participatory design facilitated a negotiation on the landscape effects of such knowledge-so far only in GRAF's hands-via agreeing on a wildfire strategy that incorporates social priorities about landscape. In turn, the fact that citizens reported to have acquired some knowledge on wildfire and wildfire management from GRAF and other agencies suggests that expert agencies can enhance the recovery of local fire knowledge through participatory planning networks. Recovering fire knowledge by re-incorporating fire use in agricultural or grazing activities of local communities is at the heart of calls for innovative wildfire management [59]. This is however a challenging task given that traditional fire knowledge systems declined or vanished with industrialization [21]. Indeed, our process took place in a metropolitan setting where local communities have very limited experience in fire management. In the Barcelona metropolitan region and in many urbanized fire-prone regions of western societies, the Fire Department's wildfire specialist groups can therefore play a key role as revitalizers of fire knowledge. Thus, a better understanding of how planning networks reconfigure traditional fire knowledge and how this reconfiguration affects community resilience and power relationships should be the object of further research [60].

Finally, our approach situates participatory wildfire planning networks in the broader framework of transformations to sustainability. Various papers have argued that addressing sustainability challenges requires a deliberate transformation of the systems that perpetuate environmental problems-minor adjustments in governance or institutions will simply not suffice [31-34,61,62]. Similarly, the attempt to coexist with wildfire has been shown to entail a radical social-ecological transformation of land-uses, settlement patterns, energy supply systems and social values about wildfires [1]. Our pilot process reveals that legitimacy and a suitable management of conflict/consensus are two key aspects underpinning the transformative capacity of a participatory wildfire planning network. The mechanisms by which our process built legitimacy among actors made possible the successful articulation of heterogeneous interests into a common purpose, i.e. informing the strategies of the Fire Department. This entailed opening the decisions shaping wildfires and landscapes from the techno-managerial up to the political realm. However, the unsuccessful articulation of actors with the potential to re-frame the wildfire issue also revealed the network's inability to rethink wildfire resilience from a management problem to a political process whereby more sustainable social-ecological configurations are *deliberately produced* [30,63]. This suggests that the debate about the ideal distribution of responsibility in risk management across governments, communities and the private sector [64] should be expanded to encompass the distribution of transformative agency across social actors. By shedding light on the political-ecological processes in which planning networks are embedded and on their transformative potential, these insights can complement new planning approaches aimed at integrating the biophysical and social dimensions of wildfire risk [4,10,11,59]. Our method can also be used to operationalize fire restoration calls [24,25,65–67] by agreeing upon which wildfires will be left to burn, under what circumstances and according to whose values within the affected communities. Our experience suggests that the effectiveness of both endeavours will depend on their ability to build legitimacy across scales and to channel social and political conflict into transformative avenues.

## 6. Conclusions

In this paper we presented a method to democratize wildfire strategies by incorporating social values about landscape in suppression and prevention planning. Based on a pilot project, we described how this can be operationalized by developing a network of researchers, practitioners and citizens across spatial and governance scales. We showed that such networks have the potential to build resilient landscapes as they facilitate discussions and agreements on strategies that reduce expected wildfire intensity and minimize the loss of social values. The outputs of these networks can likewise contribute to improve wildfire risk maps by integrating social priorities with biophysical data.

Our paper pointed at several challenges for the successful implementation of such networks, and suggested ways to address them. Practical challenges include how to develop more inclusive social valuation methods and how to overcome institutional obstacles to long-term engagement in trans-disciplinary projects. Deeper challenges were shown to be related to the risk levels derived from the region's current social-ecological setup. A trade-off between high wildfire risk and democratic wildfire management was suggested. In turn, the political negotiation on the landscape effects of expert knowledge was shown as a potentially transformative pathway that reduces risk, fosters re-learning on fire, and re-defines agency over landscape.

Our experience thus illustrates how democratizing wildfire strategies ultimately entails co-shaping the landscapes and societies of the future. The prefix "co-"refers both to a human collective involved in such deliberations and to wildfire itself. Given current risk levels, how to turn the latter from an enemy to an ally in the production of a new social-ecological setup should be given utmost attention by researchers, practitioners and policy-makers. Our method could eventually be used to prevent tragic wildfires while addressing–and correcting–their underlying social-ecological causes. However, as similar experiences are conducted beyond the pilot level and deeply antagonistic interests become involved, the challenges identified here will become ever more relevant. We therefor recommend that the projects' goals, biases, design criteria, methodological choices and public prioritizations are kept open to successive phases of monitoring and deliberation.

## **Supporting information**

**S1 Table. Timeline of the participatory process.** (DOCX)

## Acknowledgments

This paper is dedicated to the memory of Paco Muñoz, from the Montseny Association of Forest Landowners. We would like to thank all the people from government agencies, town councils, civil society organizations, museums and universities that were involved in the project for their engagement and supply of precious information. We also thank the participants that attended the public exhibitions for their contribution to the project. We acknowledge the involvement of Ana Martín and Margo Bau Collin in mapping alternative projects and the assistance of Stella Demonte and Javier Otero in the organization of the first exhibition. Isabel Lopera from ICTA-UAB contributed to the dissemination of the project 's exhibitions and results to the media. Mar Grau Satorras and Jaime García Márquez provided GIS support for Fig 1A and 1B. Jaime García Márquez also helped in the design of the landscape valuation method. Marta Borrós Vendrell helped with the layout of figures. The feedback from Tobias Kruger and two anonymous reviewers on an earlier draft of this paper are greatly appreciated.

## **Author Contributions**

Conceptualization: Iago Otero, Marc Castellnou, Itziar González.

Formal analysis: Iago Otero, Marc Castellnou, Etel Arilla, Llorenç Castell, Jordi Castellví.

Funding acquisition: Iago Otero.

Methodology: Iago Otero, Marc Castellnou, Itziar González, Etel Arilla, Llorenç Castell, Jordi Castellví.

Project administration: Iago Otero, Jonas Ø. Nielsen.

Supervision: Iago Otero, Marc Castellnou, Itziar González, Jonas Ø. Nielsen.

Validation: Iago Otero, Marc Castellnou, Itziar González.

- Visualization: Iago Otero, Itziar González, Etel Arilla, Llorenç Castell, Jordi Castellví, Francesc Sánchez.
- Writing original draft: Iago Otero.

Writing – review & editing: Marc Castellnou, Itziar González, Llorenç Castell, Jordi Castellví, Jonas Ø. Nielsen.

### References

1. Otero I, Nielsen JØ. Coexisting with wildfire? Achievements and challenges for a radical social-ecological transformation in Catalonia (Spain). Geoforum. 2017; 85:234–246.

- 2. Butler WH, Goldstein BE. The US Fire Learning Network: Springing a rigidity trap through multiscalar collaborative networks. Ecol Soc. 2010; 15(3):21.
- 3. Almstedt Å, Reed MG. Introducing a framework for good and adaptive governance: An application to fire management planning in Canada's boreal forest. For Chron. 2013; 89(05):664–74.
- Plana E, Martín D, Font M, Serra M, Molina D. Social factor and territorial dimension of wildfire risk management: managing societal involvement and cross-sectoral planning. FIREfficient Project; 2015. Report No.: 15.
- Gazzard R, McMorrow J, Aylen J. Wildfire policy and management in England: an evolving response from Fire and Rescue Services, forestry and cross-sector groups. Phil Trans R Soc B. 2016; 371 (1696):20150341. https://doi.org/10.1098/rstb.2015.0341 PMID: 27216511
- Chapin FS, Trainor SF, Huntington O, Lovecraft AL, Zavaleta E, Natcher DC, et al. Increasing Wildfire in Alaska's Boreal Forest: Pathways to Potential Solutions of a Wicked Problem. BioScience. 2008; 58 (6):531–40.
- 7. Everett Y, Fuller M. Fire Safe Councils in the Interface. Soc Nat Resour. 2011; 24(4):319–33.
- Moore P, Ganz D, Tan LC, Enters T, Durst PB, editors. Communities in flames: proceedings of an international conference on community involvement in fire management. Bangkok: FAO and FireFight South East Asia; 2002.
- Morehouse BJ, O'Brien S. Facilitating Public Involvement in Strategic Planning for Wildland Fire Management. Prof Geogr. 2008; 60(4):495–507.
- Ager AA, Kline JD, Fischer AP. Coupling the Biophysical and Social Dimensions of Wildfire Risk to Improve Wildfire Mitigation Planning. Risk Anal. 2015; 35(8):1393–406. <u>https://doi.org/10.1111/risa.</u> 12373 PMID: 25968881
- Fischer AP, Spies TA, Steelman TA, Moseley C, Johnson BR, Bailey JD, et al. Wildfire risk as a socioecological pathology. Front Ecol Environ. 2016; 14(5):276–84.
- Folke C, Hahn T, Olsson P, Norberg J. Adaptive governance of social-ecological systems. Annu Rev Environ Resour. 2005; 30(1):441–73.
- Cash D, Adger WN, Berkes F, Garden P, Lebel L, Olsson P, et al. Scale and cross-scale dynamics: governance and information in a multilevel world. Ecol Soc. 2006; 11(2).
- 14. Olsson P, Folke C, Galaz V, Hahn T, Schultz L. Enhancing the fit through adaptive co-management: creating and maintaining bridging functions for matching scales in the Kristianstads Vattenrike Biosphere Reserve, Sweden. Ecol Soc. 2007; 12(1).
- Berkes F. Community-based conservation in a globalized world. Proc Natl Acad Sci. 2007; 104 (39):15188–93. https://doi.org/10.1073/pnas.0702098104 PMID: 17881580
- Everett Y. Community participation in fire management planning: a case from California, USA. In: Moore P, Ganz D, Tan LC, Enters T, Durst PB, editors. Communities in flames: proceedings of an international conference on community involvement in fire management. Bangkok: FAO and FireFight South East Asia; 2002. pp. 125–33.
- Tàbara D, Saurí D, Cerdan R. Forest fire risk management and public participation in changing socioenvironmental conditions: a case study in a Mediterranean region. Risk Anal. 2003; 23(2):249–60. PMID: 12731810
- Morehouse BJ, O'Brien S, Christopherson G, Johnson P. Integrating values and risk perceptions into a decision support system. Int J Wildland Fire. 2010; 19(1):123.
- Rawluk A, Ford RM, Neolaka FL, Williams KJ. Public values for integration in natural disaster management and planning: A case study from Victoria, Australia. J Environ Manage. 2017; 185:11–20. <u>https://doi.org/10.1016/j.jenvman.2016.10.052</u> PMID: 27810108
- González-Hidalgo M, Otero I, Kallis G. Seeing beyond the smoke: the political ecology of fire in Horta de Sant Joan (Catalonia). Environ Plan A. 2014; 46(5):1014–1031.
- Huffman MR. The many elements of Traditional Fire Knowledge: synthesis, classification, and aids to cross-cultural problem solving in fire-dependent systems around the world. Ecol Soc. 2013; 18(4). https://doi.org/10.5751/ES-06288-180472
- 22. Castellnou M, Miralles M, Molina D. Estrategia, tácticas y maniobras en incendios forestales [Internet]. Máster Fuego, University of Lleida; 2013. Available from: https://www.researchgate.net/profile/ Domingo\_Molina\_Terren/publication/301643655\_ESTRATEGIA\_TACTICAS\_Y\_MANIOBRAS\_en\_ Incendios\_Forestales/links/571f6ef108aead26e71b6312/ESTRATEGIA-TACTICAS-Y-MANIOBRASen-Incendios-Forestales.pdf
- 23. Campbell D. The Campbell Prediction System: a wild land fire prediction system & language. 2nd ed. D. Campbell; 1995.

- 24. Donovan GH, Brown TC. Be careful what you wish for: the legacy of Smokey Bear. Front Ecol Environ. 2007 Mar 1; 5(2):73–9.
- Calkin DE, Thompson MP, Finney MA. Negative consequences of positive feedbacks in US wildfire management. For Ecosyst. 2015 Apr 14; 2(1):9.
- Moritz MA, Batllori E, Bradstock RA, Gill AM, Handmer J, Hessburg PF, et al. Learning to coexist with wildfire. Nature. 2014 Nov 6; 515(7525):58–66. https://doi.org/10.1038/nature13946 PMID: 25373675
- 27. Castellnou M. Talk at the International Congress on Prescribed Fires. Barcelona, 1–3 February 2017 [Internet]. Available from: https://www.youtube.com/watch?v=vrnHo6LAj8o
- 28. Robbins P. Political Ecology: a critical introduction. 2nd ed. Wiley-Blackwell; 2012.
- **29.** Martinez-Alier J. The environmentalism of the poor: a study of ecological conflicts and valuation. Cheltenham: Edward Elgar; 2002.
- Swyngedouw E, Heynen NC. Urban Political Ecology, Justice and the Politics of Scale. Antipode. 2003 Nov 1; 35(5):898–918.
- O'Brien K. Global environmental change II: From adaptation to deliberate transformation. Prog Hum Geogr. 2012; 36(5).
- O'Brien K. Global environmental change III Closing the gap between knowledge and action. Prog Hum Geogr. 2013 Aug 1; 37(4):587–96.
- 33. Moser SC. Editorial overview: Transformations and co-design: Co-designing research projects on social transformations to sustainability. Curr Opin Environ Sustain. 2016; 20:v–viii.
- Brown K, O'Neill S, Fabricius C. Social science understandings of transformation. In: ISSC/UNESCO World Social Science Report 2013: Changing Global Environments. OECD Publishing, Paris/Unesco Publishing; 2013. pp. 100–106.
- **35.** Reason P, Bradbury H. The SAGE handbook of action research. Participative inquiry and practice. SAGE Publications; 2008.
- Herr K, Anderson G. The action research dissertation. A guide for students and faculty. SAGE Publications; 2005.
- Otero I, Marull J, Tello E, Diana GL, Pons M, Coll F, et al. Land abandonment, landscape, and biodiversity: questioning the restorative character of the forest transition in the Mediterranean. Ecol Soc. 2015; 20(2).
- 38. Otero I, Arilla E. Prevenció d'incendis forestals i nou model territorial. Sitja del Llop. 2015; 38:22-3.
- Bartolomé J, Plaixats J, Fanlo R, Boada M. Conservation of isolated Atlantic heathlands in the Mediterranean region: effects of land-use changes in the Montseny biosphere reserve (Spain). Biol Conserv. 2005 Mar; 122(1):81–8.
- 40. Castellnou M, Pagés J, Miralles M, Piqué M. Tipificación de los incendios forestales de Cataluña. Elaboración del mapa de incendios de diseño como herramienta para la gestión forestal. In: 5º Congreso Forestal Español. Montes y sociedad: saber qué hacer. Ávila: Sociedad Española de Ciencias Forestales and Junta de Castilla y León; 2009.
- **41.** Costa P, Castellnou M, Larrañaga A, Miralles M, Kraus PD. Prevention of large wildfires using the fire types concept. Handbook prepared for the project Fire Paradox under the EU Sixth Framework Programme. Generalitat de Catalunya, Departament d'Interior, Unitat Tècnica GRAF; 2011.
- 42. Vicente O, Nebot E. Incendi de Gualba i Santa Coloma de Farners– 10 d'agost de 1994 [Wildfire report]. Generalitat de Catalunya, Departament d'Interior, Direcció General de Prevenció, Extinció d'Incendis i Salvaments, Subdirecció General Operativa, Divisió de Grups Operatius Especials; 2012.
- 43. Conflicting fire regimes and social-ecological projects. The political ecology of wildfires in Catalonia [Internet]. Available from: https://www.iri-thesys.org/research/research-groups/urben\_rural/conflicting-fire-regimes/conflicting\_fire\_regimes
- **44.** Robinson J. Being undisciplined: transgressions and intersections in academia and beyond. Futures. 2008; 40:70–86.
- Asara V, Muraca B. Indignados (Occupy!). In: D'Alisa G, Demaria F, Kallis G, editors. Degrowth A vocabulary for a new era. Routledge; 2015. pp. 169–71.
- Abrams JB, Knapp M, Paveglio TB, Ellison A, Moseley C, Nielsen-Pincus M, et al. Re-envisioning community-wildfire relations in the U.S. West as adaptive governance. Ecol Soc. 2015; 20(3).
- **47.** Howitt R. Coexisting with Fire? A Commentary on the Scale Politics of Adaptation. Geogr Res. 2014 Feb 1; 52(1):61–4.
- **48.** O'Neill SJ, Handmer J. Responding to bushfire risk: the need for transformative adaptation. Environ Res Lett. 2012; 7(1):014018.

- 49. Buizer M, Kurz T. Too hot to handle: Depoliticisation and the discourse of ecological modernisation in fire management debates. Geoforum. 2016 Jan; 68:48–56.
- 50. Mouffe C. On the Political. Routledge; 2005.
- Carroll MS, Higgins LL, Cohn PJ, Burchfield J. Community Wildfire Events as a Source of Social Conflict. Rural Sociol. 2006 Jun 1; 71(2):261–80.
- Carroll MS, Cohn PJ, Seesholtz DN, Higgins LL. Fire as a Galvanizing and Fragmenting Influence on Communities: The Case of the Rodeo–Chediski Fire. Soc Nat Resour. 2005 Mar 23; 18(4):301–20.
- 53. Eriksen C. Why do they burn the 'bush'? Fire, rural livelihoods, and conservation in Zambia. Geogr J. 2007 Sep 1; 173(3):242–56.
- Kull CA. Madagascar aflame: landscape burning as peasant protest, resistance, or a resource management tool? Polit Geogr. 2002 Sep; 21(7):927–53.
- Sletto B. The Knowledge that Counts: Institutional Identities, Policy Science, and the Conflict Over Fire Management in the Gran Sabana, Venezuela. World Dev. 2008 Oct; 36(10):1938–55.
- Otero I, Niewöhner J, Krueger T, Doğmuş ÖC, Himmelreich J, Sichau C, et al. The position of scientists in transformations of human-environment systems. An inquiry into IRI THESys research practices. THESys Discuss Pap. 2017;2017–1.
- Xue J. Is eco-village/urban village the future of a degrowth society? An urban planner's perspective. Ecol Econ. 2014 Sep; 105:130–8.
- Brummel RF, Nelson KC, Souter SG, Jakes PJ, Williams DR. Social learning in a policy-mandated collaboration: community wildfire protection planning in the eastern United States. J Environ Plan Manag. 2010 Sep 1; 53(6):681–99.
- Tedim F, Leone V, Xanthopoulos G. A wildfire risk management concept based on a social-ecological approach in the European Union: Fire Smart Territory. Int J Disaster Risk Reduct. 2016 Sep; 18:138– 53.
- **60.** Eriksen C, Prior T. The art of learning: wildfire, amenity migration and local environmental knowledge. Int J Wildland Fire. 2011; 20(4):612–24.
- Nelson DR, Adger WN, Brown K. Adaptation to Environmental Change: Contributions of a Resilience Framework. Annu Rev Environ Resour. 2007; 32(1):395–419.
- Asara V, Otero I, Demaria F, Corbera E. Socially sustainable degrowth as a social–ecological transformation: repoliticizing sustainability. Sustain Sci. 2015 Jul 5; 10(3):375–84.
- **63.** Otero I, Kallis G, Aguilar R, Ruiz V. Water scarcity, social power and the production of an elite suburb: The political ecology of water in Matadepera, Catalonia. Ecol Econ. 2011 May 15; 70(7):1297–308.
- 64. McLennan BJ, Handmer J. Reframing responsibility-sharing for bushfire risk management in Australia after Black Saturday. Environ Hazards. 2012 Mar 1; 11(1):1–15.
- North MP, Stephens SL, Collins BM, Agee JK, Aplet G, Franklin JF, et al. Reform forest fire management. Science. 2015 Sep 18; 349(6254):1280–1. https://doi.org/10.1126/science.aab2356 PMID: 26383934
- Thompson M, Dunn C, Calkin D. Wildfires: Systemic changes required. Science. 2015 Nov 20; 350 (6263):920–920.
- Regos A, Aquilué N, Retana J, De Cáceres M, Brotons L. Using Unplanned Fires to Help Suppressing Future Large Fires in Mediterranean Forests. PLoS ONE. 2014 Apr 11; 9(4):e94906. <u>https://doi.org/10.1371/journal.pone.0094906</u> PMID: 24727853